

4CB4/1.2/AI7/1978

The Airline Skier: 1977-78 Season

JAN 1 5 1978

A Comparison of the Skiers Traveling by Commercial Air in Five Skier Studies Conducted at Aspen, Vail, Steamboat, Winter Park and Copper Mountain

Business Research Division • Graduate School of Business Administration University of Colorado • Boulder, Colorado

THE AIRLINE SKIER: 1977-78 SEASON

Ŭ.

A Comparison of the Skiers Traveling by Commercial Air in Five Skier Studies Conducted at Aspen, Vail, Steamboat, Winter Park and Copper Mountain"

by

C. R. Goeldner

Business Research Division Graduate School of Business Administration University of Colorado Boulder, Colorado

Price Per Copy - \$15.00

Additional copies may be ordered from:

Business Research Division Grduate School of Business Administration University of Colorado Boulder, Colorado 80309

ISBN No. 0-89478-045-X

Copyright 1978

By the Business Research Division Graduate School of Business Adminstration University of Colorado Boulder, Colorado 80309

TABLE OF CONTENTS

	List of Tables
I.	Introduction
II.	Commercial Airline Skier Characteristics
III.	Commercial Air Transportation Characteristics
IV.	Commercial Air Accommodation Characteristics
V.	Commercial Air Ski Vacation Characteristics
	Appendix A - Questionnaire

LIST OF TABLES

Tabl	Page	2
1	Origin of Skiers, Commercial Air	
2	Commercial Airline Skiers by Geographic Region	
3	Sex	
4	Marital Status	
5	Age	
6	Age Distribution, 1974 Airline Skier Survey	
7	Education	
8	Educational Attainment, 1974 Airline Skier Survey	
9	Household Income	
10	Annual Household Income, 1974 Airline Skier Survey	
11	Occupation	
12	Occupation, 1974 Airline Skier Survey 21	
13	Skiing Ability	
14	Classification As A Skier, 1974 Airline Skier Survey	

Tab1	<u>e</u>	Ī	Page
15	Skiing Experience	•	23
16	Number of Years Skied, 1974 Airline Skier Survey	•	24
17	Party Size	•	25
18	Number of Persons in Party, 1974 Airline Skier Survey		25
19	Airlines Used	•	27
20	Mode of Transportation to Ski Area	•	28
21	Transportation From Airport, 1974 Airline Skier Survey	•	29
22	Using a Car	•	29
23	Do You Think A Car Is Necessary?	•	30
24	Lodging Accommodations	•	31
25	Type of Accommodations, 1974 Airline Skier Survey	•	32
26	Advance Lodging Reservations	•	33
27	Nights Spent At Ski Resort	•	34
28	Length of Visit, 1974 Airline Skier Survey	•	35
29	On A Package Plan	•	36
30	Number Of Ski Vacations		37

iii

Tab1	e		Pag	;e
31	Number of Ski Vacations Taken Per Year, 1974 Airline Skier Survey	•	38	}
32	Brochure Seen	•	38	;
33	Sources of Planning Information	•	39)
34	Contact In Making Travel Arrangements	•	40)
35	First Ski Vacation		41	L

INTRODUCTION

This report presents a comparison of the answers given by respondents traveling by commercial air to questions asked in five ski studies conducted at Colorado ski areas during the 1977-78 season. The areas are Aspen, Vail, Winter Park, Steamboat and Copper Mountain. During the 1977-78 season, the Aspen ski areas of Aspen Highlands, Aspen Mountain, Buttermilk and Snowmass, had 63.4 percent of their out-ofstate respondents arrive by commercial air. Vail had 62.5 percent, Winter Park had 36.6 percent, Steamboat had 47.9 percent, and Copper Mountain had 27.9 percent. The results of over 1,400 interviews are reported and provide valuable information on this important market segment of the Colorado ski industry. During the 1973-74 ski season the Business Research Division, University of Colorado, conducted a pilot study at Stapleton International Airport with eight cooperating airlines. This pilot study contained the results of interviews with 981 skiers who departed from Colorado during early March and was titled <u>The Airline Skier</u>. Where possible, comparisons will be made with the 1974 study and the current studies.

The Business Research Division at the University of Colorado has been working for a number of years to get ski areas to conduct studies using the same methodology and standardized questions so the results

I

will produce uniform comparable data. The advantage of this approach is that it permits studies to be compared and develops a base of information from which trends can be traced. The efforts have been moderately successful and one of the purposes of this report is to show the wisdom of the approach by presenting a comparison of the commercial air market segment of the five studies that were conducted in cooperation with the Business Research Division in the 1977-78 ski season. Another comparison of the in-state and out-of-state skiers has been published in another report titled <u>The Colorado Skier: 1977-78</u> Season.

While the studies compared in this report did not all use the same methodology (four used the lift interview technique and one the ticket office mail-back questionnaire technique), nor were all of the questions standardized, enough were identical or similar to be reported here. The reader will note in some cases question categories had to be collapsed for comparison purposes because standardized questions were not used. The reader will be able to spot cases where standard questions and categories were not used and the problems this created.

A brief review of the methodology used in each of the studies follows. The purpose of this description is to allow the reader to determine the comparability of the studies and the validity of the comparisons.

Aspen

The Aspen study used the ski lift interview technique. Respondents were selected in a manner that required interviews to be conducted on each mountain in the proportion to which skiers skied the area in

the 1975-76 season, which was considered to be the last normal season. The interviews were conducted by volunteers recruited and supervised by the Aspen Chamber of Commerce. The sampling plan generated 848 successfully completed interviews over the time period of December 27 to March 31, of which 505 were with commercial air respondents. The research design called for respondents to be 18 years of age or over and to reside outside of Pitkin County.

Vail

The ski lift interview technique was also used at Vail. A sampling plan was developed which generated 911 successfully completed interviews on sample days distributed over the entire ski season, of which 363 were with commercial air respondents. Interviewers were hired by Vail Associates to conduct the interviews and the study was supervised by their marketing department. Local skiers were included in this study.

Winter Park

The ski lift interview technique was also used at Winter Park. The sampling plan developed generated 902 successfully completed interviews during 32 sample days spread throughout the heart of the 1977-78 operating season. A total of 210 interviews were conducted with commercial air respondents. Interviewing took place on sample days from December 22 through April 1. All interviewing was conducted by Winter Park personnel.

Steamboat

Steamboat also used the lift interview technique and had the most rigorous sample design. Steamboat interviewed every day of the year in proportion to the number of skiers that typically came on that day based on past records. This sample generated over 1,000 successfully completed interviews, of which 337 were with commercial air respondents. All interviewers were hired, trained and controlled by Steamboat. Local skiers were included in this study.

Copper Mountain

Copper Mountain used a different methodology from the other areas in conducting its survey--the self-administered questionnaire technique. Approximately every 1,000th person who approached the ticket window was asked to participate in their two-part survey. Part One was a short questionnaire respondents were asked to fill out and return to the ticket window; and Part Two was a lengthy follow-up questionnaire mailed to respondents after they returned home. This methodology also created a bias toward male skiers as they tend to purchase the lift tickets for the party. Since this methodology has been utilized previously by Copper Mountain, they continue to use it so their studies will be comparable.

Comment

While all of the studies used sound research methodologies, they were each slightly different; consequently the reader should use the information carefully. Data reported in the survey are raw data

from the computer printouts and no attempt has been made to weight the data by sample size or skier visits. Results of each of the five studies are reported and a total arrived at by simple addition.

Organization

The report is divided into five chapters. Chapter I presents the introduction; Chapter II presents the general characteristics of commercial air skier respondents; Chapter III covers transportation characteristics; Chapter IV deals with accommodation information; and Chapter V contains ski vacation characteristics.

COMMERCIAL AIRLINE SKIER CHARACTERISTICS

Origin

Of the 1,459 respondents using commercial air included in the five surveys conducted at Aspen, Vail, Winter Park, Steamboat and Copper Mountain, the largest group came from California (11.5 percent), as shown in Table 1. While California was the top-ranked state, sending 168 of the 1,459 respondents, Texas was close behind with 140 (9.6 percent), followed by Illinois with 138 (9.5 percent). New York with 7.1 percent was the only other state with over 5 percent of the respondents. Florida was the fifth ranked state with 70 respondents, or 4.8 percent of the total. These five states accounted for 42.5 percent of the commercial air respondents. The United States was the residence of 94.3 percent of the respondents. Canada was the major foreign residence of those interviewed, providing 40 (2.7 percent) of the 1,457 respondents. Other foreign countries added 28 (1.9 percent), and Mexico contributed 16 (1.1 percent).

Skiers from California represented the largest share of Vail's market, totaling 13.5 percent, and of Aspen's market, totaling 12.3 percent.

6

II

TUDDUD T	T_{I}	ABL	E	1	
----------	---------	-----	---	---	--

					Win	nter			Coj	pper		
	A	spen	Va	ail	Pa	ark	Stea	amboat	Mour	ntain	Tot	
	#	%	#	%	#	%	#	%	#	%	#	%
Alabama	5	1.0%	2	0.6%	1	0.5%	_	- %	1	2.3%	9	0.6%
Alaska	1	0.2	-	_	_	-	1	0.3	-	_	2	0.1
Arizona	4	0.8	4	1.1	_	-	3	0.9	_	-	11	0.8
Arkansas	5	1.0	1	0.3	1	0.5	-	_	-	-	7	0.5
California	62	12.3	49	13.5	11	5.2	41	12.2	5	11.4	168	11.5
Colorado	-	-			_	-	-		1	2.3	1	0.1
Connecticut	13	2.6	8	2.2	5	2.4	12	3.6	_	-	38	2.6
Delaware	1	0.2	1	0.3	_	_	1	0.3		-	3	0.2
District of Columbia	10	2.0	6	1.7	3	1.4	8	2.4	-	-	27	1.9
Florida	28	5.5	14	3.9	7	3.3	17	5.0	4	9.1	70	4.8
Georgia	21	4.2	14	3.9	5	2.4	8	2.4	1	2.3	49	3.4
Hawaii	5	1.0	3	0.8	1	0.5	4	1.2	-		13	0.9
Idaho	1	0.2	_	-	1	0.5	-	-	-	-	2	0.1
Illinois	51	10.1	31	8.5	22	10.5	30	8.9	4	9.1	138	9.5
Indiana	7	1.4	4	1.1	3	1.4	4	1.2	1	2.3	19	1.3
Iowa	1	0.2	2	0.6	2	1.0	3	0.9	-	-	8	0.6
Kansas	3	0.6	7	1.9	3	1.4	1	0.3	1	2.3	15	1.0
Kentucky	-	-	-	-	-	-	1	0.3	1	2.3	2	0.1
Louisiana	5	1.0	6	1.7	15	7.1	11	3.3	-	-	37	2.5
Maine	-	-	1	0.3	_	-	-	-	-	—	1	0.1
Maryland	6	1.2	4	1.1	2	1.0	4	1.2	1	2.3	17	1.2
Massachusetts	11	2.2	7	1.9	-	—	10	3.0	-	-	28	1.9
Michigan	27	5.3	9	2.5	9	4.3	14	4.2	1	2.3	60	4.1
Minnesota	6	1.2	12	3.3	8	3.8	20	5.9	1	2.3	47	3.2
Mississippi	3	0.6	1	0.3	3	1.4	1	0.3	-	-	8	0.6
Missouri	7	1.4	11	3.0	33	15.7 -	10	3.0	1	2.3	62	4.3
Montana	-	-	3	0.8	-	-	-	-	-	-	3	0.2
Nebraska	1	0.2	1	0.3	-	-	2	0.6	-	-	4	0.3
Nevada	-	-	-	-	-	-	1	0.3	-	-	1	0.1
New Hampshire	3	0.6	1	0.3	-	-	-	-	1	2.3	5	0.3
New Jersey	9	1.8	6	1.7	1	0.5	4	1.2	1	2.3	21	1.4

ORIGIN OF SKIERS (COMMERCIAL AIR)

TABLE 1 (Continued)

						Wi	nter			Co	pper		
		А	spen	v	ail	P	ark	Ste	amboat	Mou	ntain	То	tal
		#	%	#	%	#	%	#	%	#	%	#	%
New Mexico	*)	-	- %	1	0.3%	_	- %	2	0.6%	_	- %	3	0.2%
New York		50	9.9	27	7.4	4	1.9	19	5.6	3	6.8	103	7.1
North Carolina		7	1.4	5	1.4	2	1.0	2	0.6	2	4.5	18	1.2
North Dakota		-	-	-	-	1	0.5	-	-	1	2.3	2	0.1
Ohio		19	3.8	18	5.0	2	1.0	8	2.4	1	2.3	48	3.3
Oklahoma		6	1.2	2	0.6	3	1.4	3	0.9		-	14	1.0
Oregon		1	0.2	4	1.1	3	1.4	3	0.9	-	-	11	0.8
Pennsylvania		21	4.2	14	3.9	2	1.0	9	2.7	-	-	46	3.2
Rhode Island		1	0.2	1	0.3	-	-	-	-	-	-	2	0.1
South Carolina		2	0.4	2	0.6		-	1	0.3	1	2.3	6	0.4
South Dakota		-	-	1	0.3	1	0.5	-	-	-	-	2	0.1
Tennessee		9	1.8	3	0.8	1	0.5	3	0.9	1	2.3	17	1.2
Texas		28	5.5	23	6.3	44	21.0	40	11.9	5	11.4	140	9.6
Utah		-	-		-	-	-	1	0.3	_	-	1	0.1
Vermont		2	0.4	2	0.6	1	0.5	1	0.3		-	6	0.4
Virginia		10	2.0	5	1.4	-	-	1	0.3	2	4.5	18	1.2
Washington		6	1.2	4	1.1	1	0.5		-	-		11	0.8
West Virginia		_	-	' 3	0.8	-	-	1	0.3	-		4	0.3
Wisconsin		15	3.0	11	3.0	6	2.9	9	2.7	2	4.5	43	3.0
Wyoming		-	-	-	_	-	-	1	0.3	-	-	1	0.1
Canada		17	3.4	9	2.5	-		13	3.9	1	2.3	40	2.7
Mexico		3	0.6	11	3.0	1	0.5	1	0.3	-	-	16	1.1
Other Foreign		12	2.4	9	2.5	2	1.0	5	1.5	-	-	28	1.9
No Response	-	-	-	-	-	-	-	3	0.8	_	-	3	0.3
Annenion and Stuff 🛓 (S. Sault (S.		505	100.4%	363	100.5%	210	100.4%	337	100.5%	44	100.4%	1,459	100.3%

ORIGIN OF SKIERS (COMMERCIAL AIR)

Winter Park secured the largest share of its market from Texas, 21.0 percent. Texas also contributed heavily to Steamboat, 11.9 percent, and to Copper Mountain, 11.4 percent.

Illinois skiers were spread fairly evenly across the five areas, accounting for 10.1 percent of Aspen, 8.5 percent at Vail, 10.5 percent at Winter Park, 8.9 percent at Steamboat, and 9.1 percent at Copper Mountain.

The top five states producing commercial air skiers at each of the areas are as follows.

Aspen	Vail	Winter Park	Steamboat	Copper Mountain
1. California 2. Illinois 3. New York 4. Texas > tie 5. Florida > tie	 California Illinois New York Texas Ohio 	 Texas Missouri Illinois Louisiana California 	 California Texas Illinois Minnesota New York 	 California > tie Texas Illinois > tie Florida New York

One would expect the air skiers and out-of-state skiers to have similar origins. This was true; however, there were some changes in the rankings. The out-of-state rankings are shown below, revealing the importance of the automobile as a transportation mode in some states.

Aspen	Vail	Winter Park	Steamboat	Copper Mountain
 California Illinois New York Texas Michigan 	 California Texas Illinois New York Ohio 	 Texas Missouri Kansas Illinois Nebraska 	 Texas Illinois California Minnesota Wyoming 	 Texas Kansas Missouri Illinois Minnesota and Nebraska

Geographic Region

The East North Central region (comprised of Illinois, Indiana, Michigan, Ohio and Wisconsin) was the home of the largest group of the 1,459 air respondents, accounting for 21.1 percent of the total. The South Atlantic region (Delaware, District of Columbia, Florida, Georgia, Maryland, North Carolina, South Carolina, West Virginia and Virginia) contributed 14.5 percent of the total, and the Pacific region (Alaska, California, Hawaii, Oregon and Washington) contributed 14.1 percent. See Table 2.

Skiers coming from the East North Central region had the greatest representation at the Aspen ski area, accounting for 23.6 percent, and Copper Mountain, 20.5 percent.

South Atlantic skiers accounted for 25.0 percent of Copper Mountain's air business and 16.9 percent of Aspen's. Representation from the Pacific region showed the heaviest concentration at Vail (16.5 percent) and Aspen (14.9 percent). Geographic region leaders are shown below:

- 1. Mountain: Steamboat leads with 2.4 percent.
- 2. East North Central: Aspen leads with 23.6 percent; Vail and Winter Park tie for second.
- 3. West North Central: Winter Park leads with 22.9 percent; Steamboat is second.
- 4. Middle Atlantic: Aspen leads with 15.9 percent; Vail is second.
- 5. West-South Central: Winter Park predominates with 30 percent; Steamboat is second.
- 6. South Atlantic: Copper Mountain leads with 25.0 percent; Aspen is second with 16.9 percent.
- 7. Pacific: Vail leads with 16.5 percent, followed by Aspen with 14.9 percent.
- 8. New England: Steamboat leads with 6.9 percent, followed by Aspen with 6.0 percent.
- 9. East South Central: Copper Mountain leads with 6.9 percent.
- 10. Canada: Steamboat has 3.4 percent; Aspen has 3.4 percent.
- 11. Mexico: Vail predominates with 3.0 percent.
- 12. Other Foreign: Aspen predominates with 2.4 percent.

						ter			_	oper	m - 4	
		spen		ail		irk		amboat		ntain	Tot	
	#	%	#	%	#	%	#	%	#	%	#	%
Mountain	5	1.0%	8	2.2%	1	0.5%	8	2.4%	1	2.3%	23	1.6%
Arizona	4	0.8	4	1.1	-	- H.	3	0.9	-	-	11	0.6
Idaho	1	0.2	_	-	1	0.5	_	-	-	-	2	0.1
Colorado	-	_	-	-	-	-	-	-	1	2.3	1	0.1
Montana	_	-	3	0.8	-	- <u>-</u>	_	_		_	3	0.2
Nevada	_	-	-	-		-	1	0.3		-	1	0.1
New Mexico	_	_	1	0.3	_	2.1	2	0.6	-	-	3	0.2
Utah	_	_	_	-	_	-	1	0.3	_	-	1	0.1
Wyoming	-	-	-	-	-	-	1	0.3	- T.	5.0	1	0.1
East North Central	119	23.6	73	20.1	42	20.1	65	19.4	9	20.5	308	21.1
Illinois	51	10.1	31	8.5	22	10.5	30	8.9	4	9.1	138	9.5
Indiana	7	1.4	4	1.1	3	1.4	4	1.2	1	2.3	19	1.3
Michigan	27	5.3	9	2.5	9	4.3	14	4.2	1	2.3	60	4.1
Ohio	19	3.8	18	5.0	2	1.0	8	2.4	1	2.3	48	3.3
Wisconsin	15	3.0	11	3.0	6	2.9	9	2.7	2	4.5	43	2.9
West North Central	18	3.6	34	9.4	48	22.9	36	10.7	4	9.2	140	9.6
Iowa	1	0.2	2	0.6	2	1.0	3	0.9	- 1	-	8	0.6
Kansas	3	0.6	7	1.9	3	1.4	1	0.3	1	2.3	15	1.0
North Dakota	_	1	_	0.00	1	0.5	-	-	1	2.3	2	0.1
South Dakota	_	_	1	0.3	1	0.5	_	-	-	-	2	0.1
Minnesota	6	1.2	12	3.3	8	3.8	20	5.9	1	2.3	47	3.2
Missouri	7	1.4	11	3.0	33	15.7	10	3.0	1	2.3	62	4.3
Nebraska	1	0.2	1	0.3	-	-	2	0.6	-	-	4	0.3
Middle Atlantic	80	15.9	47	13.0	7	3.4	32	9.5	4	9.1	170	11.7
New Jersey	9	1.8	6	1.7	1	0.5	4	1.2	1	2.3	21	1.4
New York	50	9.9	27	7.4	4	1.9	19	5.6	3	6.8	103	7.1
Pennsylvania	21	4.2	14	3.9	2	1.0	9	2.7	_	_	46	3.2

COMMERCIAL AIRLINE SKIERS BY GEOGRAPHIC REGION

TABLE 2

TABLE 2 (Continued)

COMMERCIAL AIRLINE SKIERS BY GEOGRAPHIC REGION

						nter	0.5			pper	Tot	- 1
	As	spen	V.	ail	₽ <i>a</i> ∦	ark%		amboat%	Mou	ntain%	#	2a1 %
		%	-₹F	%	11	/0	11	/0	11	/0	11	/0
West South Central	44	8.7%	32	8.9%	63	30.0%	54	16.1%	5	11.4%	198	13.6%
Arkansas	5	1.0	1	0.3	1	0.5	—	—	-	-	7	0.5
Louisiana	5	1.0	6	1.7	15	7.1	11	3.3	-	-	37	2.5
Oklahoma	6	1.2	2	0.6	3	1.4	3	0.9	-	-	14	1.0
Texas	28	5.5	23	6.3	44	21.0	40	11.9	5	11.4	140	9.6
South Atlantic	85	16.9	54	15.1	19	9.1	43	12.8	11	25.0	212	14.5
Delaware	1	0.2	1	0.3	-	-	1	0.3	-	-	3	0.2
District of Columbia	10	2.0	6	1.7	3	1.4	8	2.4	_	-	27	1.9
Florida	28	5.5	14	3.9	7	3.3	17	5.0	4	9.1	70	4.8
Georgia	21	4.2	14	3.9	5	2.4	8	2.4	1	2.3	49	3.4
Maryland	6	1.2	4	1.1	2	1.0	4	1.2	1	2.3	17	1.2
North Carolina	- 7	1.4	5	1.4	2	1.0	2	0.6	2	4.5	18	1.2
South Carolina	2	0.4	2	0.6	- 1	101 -	1	0.3	1	2.3	6	0.4
West Virginia	_	_	3	0.8	_	-	1	0.3	_	-	4	0.3
Virginia	10	2.0	5	1.4	-	-	1	0.3	2	4.5	18	1.2
Pacific	75	14.9	60	16.5	16	7.6	49	14.6	5	11.4	205	14.1
Alaska	1	0.2	_	-	-	-	1	0.3	-	-	2	0.1
California	62	12.3	49	13.5	11	5.2	41	12.2	5	11.4	168	11.5
Hawaii	5	1.0	3	0.8	1	0.5	4	1.2	-	-	13	0.9
Oregon	1	0.2	4	1.1	3	1.4	3	0.9	-	-	11	0.8
Washington	6	1.2	4	1.1	1	0.5	-	cala - a c		uc" - 1	11	0.8
New England	30	6.0	20	5.6	6	2.9	23	6.9	1	2.3	80	5.4
Connecticut	13	2.6	8	2.2	5	2.4	12	3.6	-	-	38	2.6
Maine		S	1	0.3	- N.	-	-	n i- bi	-1 = 0		1	0.1
Massachusetts	11	2.2	7	1.9	-	-	10	3.0	_	-	28	1.9
New Hampshire		0.6	1	0.3			-	-	1	2.3	5	0.3
Rhode Island	1	0.2	1	0.3	-	-	-	-	_	-	2	0.1
Vermont	2	0.4	2	0.6	1	0.5 -	1	0.3	-	_	6	0.4

TABLE 2 (Continued)

	А	spen	v	/ail		nter ark	Ste	amboat		pper ntain	Tot	al ^a
	#	%	#	%	#	%	#	%	#	%	#	%
East South Central	17	3.4%	6	1.7%	5	2.4%	5	1.5%	3	6.9%	36	2.5%
Alabama	5	1.0	2	0.6	1	0.5	-	-	1	2.3	9	0.6
Kentucky	-	-	-	-	-	-	1	0.3	1	2.3	2	0.1
Mississippi	3	0.6	1	0.3	3	1.4	1	0.3	-	-	8	0.6
Tennessee	9	1.8	3	0.8	1	0.5	3	0.9	1	2.3	17	1.2
Canada	17	3.4	9	2.5	-	-	13	3.9	1	2.3	40	2.7
Mexico	3	0.6	11	3.0	1	0.5	1	0.3	-	-	16	1.1
Other Foreign	12	2.4	9	2.5	2	1.0	5	1.5	-	-	28	1.9
No Response	505	100.4%	363		<u>-</u> 210	-	<u>3</u> 337	<u>0.9</u> 100.5%		 100.4%	<u>3</u> 1,459	0.1

COMMERCIAL AIRLINE SKIERS BY GEOGRAPHIC REGION

^a Totals may not add due to rounding.

Sex

Table 3 shows that the commercial air skiers split approximately 60-40 between male and female. As mentioned in Chapter I, the differences in the Copper Mountain figures can be explained by the sampling method. A comparison of these figures with those collected in 1974 in a pilot study of skiers using air transportation shows a slight increase in females. The 1974 figures were male--66.6 percent, female-32.5 percent, and no response--0.9 percent. Winter Park and Aspen showed the highest percentage of females.

TABLE	3
SEX	

er an an Anar	A	spen	v	ail		nter ark	Ste	amboat		pper ntain	То	tal
	#	%	#	%	#	%	#	%	#	%	#	%
Male	290	57.4%	234	64.5%	120	57.1%	213	63.2%	36	81.8%	893	61.2%
Female	212	42.0	129	35.5	89	42.4	124	36.8	8	18.2	562	38.5
No Response	3	0.6	-	-	1	0.5	_	_		_	4	0.3
e tur e e e le la	505	100.0%	363	100.0%	210	100.0%	337	100.0%	44	100.0%	1,459	100.0%

Marital Status

Commercial air skiers recorded a slight edge to those married (50.1 percent) over those single (48.9 percent), as shown in Table 4. This characteristic differs with the other market segments--the out-of-state skier and the in-state skier where the single person dominated at all five areas. In the 1974 study, 59.7 percent were married, 39.0 percent were single, and 1.2 percent did not respond. It appears that the population trend of increasing singles is also showing up on the ski slopes.

TABLE 4

MARITAL	STATUS
THATTHE	1 DIUTOD

	,	Aspen Vail		Winter Aspen Vail Park Ste		Copper Steamboat Mountain			Total				
		#	%	#	%	#	%	#	%	#	%	#	%
Married Not Married		254 248	50.3% 49.1	188 172	51.8% 47.4	113 92	53.8% 43.8	152 182	45.1% 54.0	24 20	54.5% 45.5	731 714	50.1% 48.9
No Response		<u>3</u> 505	<u>0.6</u> 100.0%	$\frac{3}{363}$	$\frac{0.8}{100.0\%}$	$\frac{5}{210}$	$\frac{2.4}{100.0\%}$	$\frac{3}{337}$	$\frac{0.9}{100.0\%}$		100.0%	14 1,459	$\frac{1.0}{100.0\%}$

Nearly one-third of the air respondents fell between the ages of 30 and 39 years, although this concentration was not consistent among the five areas. Air respondents showed older patterns than the outof-state skiers and in-state skiers. Vail had the largest representation of skiers 40 or over with 30.0 percent. The 25-29 age category showed significant representation at all five areas and was the largest group encountered at Copper Mountain (38.6 percent). Comparisons among the five areas can be seen in Table 5.

TABLE 5

AGE

					Wi	nter			Co	pper		
	Aspen Vail		ail	P	ark	rk Steamboat		Mountain		Total		
	#	%	#	%	#	%	#	%	#	%	#	%
Less than 12 years	_	- %	2	0.6%	-	- %	1	0.3%	1	2.3%	4	0.3%
13-17 years	2	0.4	16	4.4	8	3.8	17	5.0	1	2.3	44	3.0
18-19 years	14	2.8	13	3.6	9	4.3	20	5.9	6	13.6	59	4.0
20-24 years	60	11.9	45	12.4	30	14.3	60	17.8			198	13.6
25-29 years	119	23.6	84	23.1	52	24.8	70	20.8	17	38.6	342	23.4
30-39 years	188	37.2	93	25.6	52	24.8	109	32.3	11	25.0	453	31.1
40-49 years	89	17.6	81	22.3	42	20.0	46	13.6	1	2.3	259	17.8
50-59 years	26	5.1	25	6.9	14	6.7	11	3.3	6	13.6	82	5.6
60 years and over	6	1.2	3	0.8	2	1.0	3	0.9	-		14	1.0
No response	1	0.2	1	0.3	1	0.5	-	_	1	2.3	4	0.3
no responde	505	100.0%	363	100.0%	210	100.2%	337	99.9%	44	100.0%	1,459	100.1%

Age

A comparison with the 1974 airline skier study shows similar results. The 30-39 age group was the largest single category with 34.7 percent. In the 1974 study, 73.5 percent were under 40 years of age; in this study 75.4 percent were under 40. See Table 6.

TABLE 6

AGE DISTRIBUTION, 1974 AIRLINE SKIER SURVEY

	Number	Percent
Under 18	36	3.7%
18-21 years	87	8.9
22-29 years	257	26.2
30-39 years	340	34.7
40-49 years	180	18.3
50 or over	69	7.0
No response	12	1.2
	981	100.0%

Education

While education characteristics were high among both in-state and out-of-state skiers, the air skier had the highest educational levels of all. Four years of college had been completed by 33.1 percent of the respondents, while 15.1 percent had done post-graduate work and 19.7 percent had received advanced degrees. Consequently, 67.9 percent of the commercial air respondents had completed college. Another 20.9 percent had some college training, as shown in Table 7. Vail and Aspen show the most highly educated skiers with 74.9 and 72.1 percent of their skiers having completed college.

EDUCATION

					Wi	nter			Co	pper		
	A	Aspen Vail		P	Park Steamboat			Mountain		То	Total	
	#	%	#	%	#	%	#	%	#	%	#	%
0-8th grade	2	0.4%	2	0.6%	1	0.5%	8	2.4%	1	2.3%	14	1.0%
Some high school	7	1.4	19	5.2	7	3.3	15	4.5	-	-	48	3.3
High school graduate	40	7.9	14	3.9	18	8.6	21	6.2	4	9.1	97	6.7
Some college	91	18.0	56	15.4	57	27.1	93	27.6	8	18.2	305	20.9
College graduate	161	31.9	126	34.7	72	34.3	110	32.6	14	31.8	483	33.1
Post graduate work	89	17.6	62	17.1	29	13.8	31	9.2	9	20.5	220	15.1
Advanced degree	114	22.6	84	23.1	23	11.0	59	17.5	8	18.2	288	19.7
No response	1	0.2	-	_	3	1.5	-	-	-	-	4	0.3
•	505	100.0%	363	100.0%	210	100.1%	337	100.0%	44	100.1%	1,459	100.1%

Educational attainment recorded in the 1974 airline skier survey was similar, as shown in Table 8.

TABLE 8

EDUCATIONAL ATTAINMENT, 1974 AIRLINE SKIER SURVEY

	Number	Percent
8th grade or less	11	1.1%
Some high school	27	2.8
High school graduate	93	9.5
Some college	205	20.9
College graduate	286	29.2
Post graduate study	118	12.0
Advanced degree	217	22.1
No response	24	2.4
<u>ـ</u>	981	100.0%

Income

Household income is shown in Table 9 and parallels the education results. Commercial air respondents report higher incomes than either out-of-state or in-state skier groups. Over one-fourth (28.9 percent) reported annual income in the \$25,000-49,999 range, while 13.2 percent reported income in the \$50,000-74,999 range, and 15.1 percent reported income in the over \$75,000 range. The Aspen and Vail respondents tended to fall in higher income ranges than the respondents from other areas.

TABLE 9

HOUSEHOLD INCOME

					Wi	nter			Co	pper		
	А	spen	Vail		Park		Ste	Steamboat		Mountain		tal
	#	%	#	%	#	%	#	%	#	%	#	%
\$0-6,999	18	3.6%	9	2.5%	9	4.3%	10	3.0%	_	_	46	3.2%
\$7,000-14,999	45	8.9	27	7.4	23	11.0	33	9.8	3	6.8	131	9.0
\$15,000-19,999	64	12.7	36	9.9	17	8.1	37	11.0	7	15.9	161	11.0
\$20,000-24,999	42	8.3	20	5.5	26	12.4	31	9.2	5	11.4	124	8.5
\$25,000-49,999	139	27.6	92	25.3	62	29.5	107	31.8	21	47.8	421	28.9
\$50,000-74,999	65	12.9	51	14.0	26	12.4	45	13.4	5	11.4	192	13.2
575,000 and over	96	19.0	74	20.4	10	4.8	38	11.3	2	4.5	220	15.1
lo response	36	7.1	54	14.9	37	17.6	36	10.7	1	2.3	164	11.2
	505	100.1%	363	99.9%	210	100.1%	337	100.2%	44	100.1%	1,459	100.1%

A comparison with the 1974 airline skier survey shows that incomes have grown considerably. Table 10 shows income data as it was reported in the 1974 study.

	Number	Pe	ercent
\$0-6,999	29	3.0%	(3.5%)
\$7,000-9,999	57	5.8	(6.9)
\$10,000-14,999	119	12.1	(14.4)
\$15,000-19,999	106	10.8	(12.8)
\$20,000-24,999	110	11.2	(13.3)
\$25,000-49,999	242	24.7	(29.3)
\$50,000 or over	164	16.7	(19.8)
Total Respondents	827	84.3%	(100.0%)
Non-respondents	154	15.7	
Total	981	100.0%	

ANNUAL HOUSEHOLD INCOME, 1974 AIRLINE SKIER SURVEY

TABLE 10

Occupation

The largest segment of commercial air skiers reported occupations in the professional and technical fields (33.5 percent). The executive-manager category represented 16.6 percent, followed by the student category with 13.5 percent. See Table 11.

OCCUPATION

	Aspen		Vail		Winter Park		Steamboat		Copper Mountain		Total	
	#	%	#	%	#	%	#	%	#	%	#	%
Executive/manager	102	20.2%	59	16.3%	28	13.3%	46	13.6%			235	16.6%
Professional/technical	176	34.9	117	32.2	69	32.9	112	33.2			474	33.5
Teacher/professor	38	7.5	26	7.2	9	4.3	17	5.0			90	6.4
Salesman/buyer	39	7.7	28	7.7	12	5.7	31	9.2			110	7.8
Government/military	6	1.2	6	1.7	6	2.9	2	0.6			20	1.4
Office work/secretary	10	2.0	14	3.9	16	7.6	9	2.7			49	3.5
Crafts/factory/mechanic	19	3.8	4	1.1	6	2.9	10	3.0			39	2.8
Homemaker	52	10.3	26	7.2	17	8.2	26	7.7			121	8.6
Student	48	9.5	55	15.2	28	13.3	60	17.8			191	13.5
Service organization	-	-	6	1.7	-	_	-	_			6	0.4
Other	15	3.0	22	6.1	19	9.0	24	7.1			80	5.7
No response	_	-	_	-	-	-		-			-	
•	505	100.1%	363	100.3%	210	100.1%	337	99.9%			1,415	100.25

Table 12 shows the occupations reported in the 1974 airline skier survey. The reader will note there is a larger occupational spread in the 1977-78 season data.

TA	BL	Ε	1	2

	Number	Percent
Executive/manager	127	12.9%
Professional/technical	491	50.1
Teacher/professor	17	1.7
Salesman/buyer	42	4.3
Government/military	9	0.9
Secretary/office work	58	5.9
Crafts/factory/mechanic	12	1.2
Homemaker	54	5.5
Student	81	8.3
Other	90	9.2
	981	100.0%

OCCUPATION, 1974 AIRLINE SKIER SURVEY

Skier Ability

Table 13 shows how the respondents rated their skiing ability. Approximately 10 percent indicated they were beginners, 25.8 percent felt they were low intermediates, 39.2 felt they were high intermediates, 16.6 percent stated they were advanced, while 7.9 percent considered themselves experts.

ТΑ	RIF	13	Ł
TU	יו ב נו		,

SKIING ABILITY

		and the second	12.20		Wi	nter			Cop	per		
	Aspen		Vail		Park		Steamboat		Mountain		Total	
	#	%	#	%	#	%	#	%	#	%	#	%
Novice	14	2.8%	5	1.4%	8	3.8%	-	- %			27	1.9%
Beginner	27	5.3	29	8.0	22	10.5	41	12.2			119	8.4
Low intermediate	117	23.2	90	24.8	55	26.2	207	61.4			365.5	25.8
High intermediate	222	44.0	149	41.0	80	38.1					554.5	39.2
Advanced	84	16.6	78	21.5	29	13.8	89	26.4			235.5	16.6
Expert	41	8.1	12	3.3	14	6.7					111.5	7.9
No response			-	-	2	1.0	-				2	0.1
-	505	100.0%	363	100.0%	210	100.1%	337	100.0%			1,415	99.9%

A comparison with the 1974 respondent is shown in Table 14. The 1977-78 season survey shows more beginners and low intermediates. The high intermediate percentages are almost identical. If one can assume new skiers are being attracted to the industry the increase in beginner skiers is a favorable

trend.

	TA	BLE	: 14
--	----	-----	------

CLASSIFICATION AS A SKIER, 1974 AIRLINE SKIER SURVEY

	Number	Percent
Novice	21	2.1%
Beginner	47	4.8
Low intermediate	201	20.5
High intermediate	390	39.8
Advanced	234	23.9
Expert	88	9.0
*	<u>88</u> 981	100.1%

Skiing Experience

Skiers traveling by commercial air tend to be experienced skiers. Over one-third have nine or more years skiing experience. The largest group (37.2 percent) reported four to eight years experience, as shown in Table 15.

TABLE 15

		2.4			Win	nter			Copp	er		
	А	spen	n Vail		Park		Steamboat		Mountain		Total	
	#	%	#	%	#	%	#	%	#	%	#	%
Under 1 year	26	5.1%			_	- %	21	6.2%			47	4.5%
One year	16	3.2			36	17.1	13	3.9			65	6.2
2-3 years	90	17.8			35	16.6	66	19.6			191	18.2
4-8 years	180	36.6			80	38.1	131	38.9			391	37.2
9 years and over	190	37.7			56	26.6	106	31.5			352	33.5
No response	3	0.6			3	1.4					6	0.6
	505	101.0%			210	99.8%	337	100.1%			1,052	100.27

SKIING EXPERIENCE

Table 16 presents the 1974 results for comparison purposes.

ГA	BI	E	1	6

	Number	Percent
1 year	87	8.9%
2 years	77	7.8
3 years	105	10.7
4 years	129	13.1
5-8 years	252	25.7
9-11 years	131	13.4
12 or more years	197	20.1
No response	3	0.3
*	981	100.0%
	701	10

NUMBER OF YEARS SKIED, 1974 AIRLINE SKIER SURVEY

Party Size

Party size information is shown in Table 17 and indicates the most popular party size is two (25.7 percent). Singles accounted for 12.1 percent and parties of four totaled 17.5 percent.

TAI	AL E	1	7
TUI		alle .	

PARTY SIZE

					Wi	nter			Cop	per			
	A	Aspen		ail	P	Park Ste		Steamboat		Mountain		Total	
	#	%	#	%	#	%	#	%	#	%	#	%	
1 person	71	14.1%	16	6.8%	34	16.2%	35	10.4%			156	12.1%	
2 people	147	29.1	69	29.5	52	24.8	63	18.7			331	25.7	
3 people	52	10.3	39	16.7	14	6.7	44	13.1			149	11.6	
4 people	80	15.8	56	23.9	31	14.8	58	17.2			225	17.5	
5 people	26	5.1	34	14.5	17	8.1	36	10.7			113	8.8	
6 people	41	8.1	16	6.8	13	6.2	20	5.9			90	7.0	
7 or more people	61	12.2	4	1.7	23	11.0	48	14.9			136	10.6	
No response	27	5.3	-		26	12.4	33	9.8			86	6.7	
a gradina di tan	505	100.0%	234	99.9%	210	100.2%	337	100.7%			1,286	100.0%	

Table 18 shows 1974 party sizes. It appears that much larger party sizes were recorded in the 1977-78 season when compared with the 1973-74 season.

ΤÆ	B	T.1	E	1	8

NUMBER OF PERSONS IN PARTY, 1974 AIRLINE SKIER SURVEY

	 Number	Percent
1 person	342	34.9%
2 people	325	33.1
3 people	80	8.2
4 people	112	11.4
5 people	32	3.3
6 people	36	3.7
7 people	7	0.7
8 people	12	1.2
9 people	_ 35	3.6
	981	100.1%

COMMERCIAL AIR TRANSPORTATION CHARACTERISTICS

Airline Used

Table 19 shows the airline used by respondents. United proved to be the most frequently used airline with its position being strongest among Aspen and Vail skiers. Continental holds second place, followed by TWA. The entry of Delta can be seen in the tabulation and helps explain the gains recorded in skier numbers from the south. The vast improvement in air service to and from southern destinations should result in continued growth in the southern market region.

AIRLINES USED

			1	Winter					Cop	per			
	А	Aspen		Vail		Park		Steamboat		Mountain		Total	
	#	%	#	%	#	%	#	%	#	%	#	%	
United	211	41.8%	147	40.5%	52	24.8%	88	26.1%			498	35.2%	
Continental	80	15.8	48	13.2	53	25.2	80	23.7			261	18.5	
Frontier	17	3.4	16	4.4	18	8.6	29	8.6			80	5.7	
TWA	77	15.2	33	9.1	24	11.4	43	12.8			177	12.5	
Western	20	4.0	18	5.0	5	2.4	28	8.3			71	5.0	
Braniff	35	6.9	31	8.5	19	9.0	19	5.6			104	7.4	
Texas International	9	1.8	8	2.2	12	5.7	19	5.6			48	3.4	
American	-	-	10	2.8	-		-	-			10	0.7	
Delta	29	5.7	11	3.0	12	5.7	11	3.3			63	4.5	
Ozark	7	1.4	3	0.8	9	4.3	5	1.5			24	1.7	
North Central	7	1.4	3	0.8	6	2.9	11	3.3			27	1.9	
Other	8	1.6	12	3.3	-		-	-			20	1.4	
No response	5	1.0	23	6.3	-	-	4	1.2			32	2.3	
A	505	100.0%	363	99.9%	210	100.0%	337	100.0%			1,415	100.2%	

Method of Transportation to Ski Resort

Respondents were asked how they traveled to the ski resort after arriving in Colorado, and their answers were extremely varied. While the largest group utilized a rental car (32.4 percent), other modes such as private cars, commercial air and bus also enjoyed high usage (see Table 20). It is interesting to note that 30.7 percent flew directly to their destination by commercial air, although this group was concentrated among Aspen (57.2 percent) and Steamboat (44.0 percent) respondents where a direct air route was available. When Rocky Mountain Airways starts flying into Avon, Vail is expected to report similar figures.

TABLE 20

MODE	OF	TRANSP	ORTATION	TO	SKI	AREA

					Wi	nter			Co	pper		
	Aspen		Vail		Park		Steamboat		Mountain		Total	
	#	%	#	%	#	%	#	%	#	%	#	%
Ski the Summit bus	-	- %	-	- %	-	- %	-	- %	4	9.1%	4	0.3%
rivate car	31	6.1	51	14.0	40	19.0	34	10.1	19	43.2	175	12.1
Rental car	111	22.0	183	50.4	69	32.9	95	28.2	14	31.8	472	32.4
Scheduled bus	27	5.3	43	11.8	54	25.7	26	7.7	2	4.5	152	10.4
Charter bus	34	6.7	31	8.5	39	18.6	32	9.5	-	-	136	9.3
Commercial air	289	57.2	11	3.0	-		148	44.0	-	-	448	30.7
rivate air	4	0.8	-	-	-	-	1	0.3	-	-	5	0.3
ther	5	1.0	14	3.9	2	1.0	-	-	1	2.3	22	1.5
lo response	4	0.8	-	-	-	-	1	0.3	4	9.1	9	0.5
xpress bus from airport	_	-	30	8.3	6	2.9	-	-	-	-	36	2.5
mpreed bud riom darport	505	99.9%	363	99.9%	210	100.1%	337	100.1%	44	100.0%	1,459	100.0%

In comparing the mode of transportation to the ski area with the 1974 study one quickly sees several major shifts in consumer behavior--the dramatic increase in those arriving by air, largely at the expense of rental cars and chartered bus. See Table 21.

	Number	Percent
Private car	108	11.0%
Scheduled bus	111	11.3
Private plane	2	0.2
Rental car	423	43.1
Chartered bus	159	16.2
Commercial air	94	9.6
Express bus	35	3.5
Other	49	5.0
	981	99.9%

TABLE 21

TRANSPORTATION FROM AIRPORT, 1974 AIRLINE SKIER SURVEY

Method of Transportation Used at Ski Area

At Aspen and Winter Park, respondents were queried on the transportation used while staying at the resort. The largest group (56.8 percent) of commercial air skiers did not use a car. The next largest group (30.4 percent) used a rental car. See Table 22.

TABLE 22

US	ING	Α	CAR	

	٨	Aspen		Winter Vail Park Ste			Stea	mboat	Cop Moun	Total		
	#	<u>%</u>	va	%	#	%	#	%	#	%	#	%
Own car	14	2.8%			33	15.7%					47	6.6%
Rental car	145	28.7			72	34.3					217	30.4
Other	35	6.9			10	4.8					45	6.3
None	$\frac{311}{505}$	61.6			<u>95</u> 210	<u>45.2</u> 100.0%					<u>406</u> 715	<u>56.8</u> 100.1%

Is a Car Necessary

Four of the surveys included the question: "Is a car necessary?" The results are shown in Table 23 and indicate that 60.9 percent felt it was not, while 34.2 percent felt it was needed. These results are the opposite of those recorded by the in-state skier where 60.2 percent felt a car was needed and 32.8 percent stated it was not necessary.

TABLE 23

			17-4	1		nter	Cto	amhaat		opper Intain	Т о	tal
	As	spen	Vai	<u> </u>	- <u> </u>	ark	DLe	amboat	MOU	mtain		Lai
		%		%	#	%	#	%		%	#	%
Yes	127	25.1%			113	53.8%	112	33.2%	23	52.3%	375	34.2%
No	350	69.3			95	45.2	209	62.0	13	29.5	667	60.9
Not applicable	27	5.3			_	-	16	4.7	-	-	35	3.2
No response	1	0.2			2	1.0			8	18.2	19	1.7
	505	99.9%			210	100.0%	337	99.9%	44	100.0%	1,096	100.0%

DO YOU THINK A CAR IS NECESSARY?

COMMERCIAL AIR ACCOMMODATION CHARACTERISTICS

IV

Type of Lodging

Lodges and condominiums were the most popular type of accommodations with 40.4 and 40.3 percent respectively. A significant 12.2 percent stayed with friends. Table 24 presents accommodation information for the areas except for Vail.

TA	BLE	21.
_ T H	DLL	24

	1				Wi	nter			Co	pper		
	A	spen	V	ail	Pa	ark	Stea	amboat_	Mou	ntain	То	tal
	#	%	#	%	#	%	#	%	#	%	#	%
Lodge, motel, hotel	200	39.6%			94	44.7%	143	42.4%	6	13.6%	443	40.4%
Condominium	197	39.0			79	37.6	139	41.2	27	61.4	442	40.3
Rented house	18	3.6			3	1.4	3	0.9	_	-	24	2.2
With friends	64	12.7			12	5.7	49	14.5	9	20.5	134	12.2
Camper		_			-	_	1	0.3		-	1	0.1
Dormitory	13	2.6			_	-	-	-	-	-	13	1.2
Other	10	2.0			2	1.0	-	-	2	4.5	14	1.3
No response	3	0.6			1	0.5	1	0.3	—		5	0.5
Not applicable	-	-			19	9.0	1	0.3	_	-	20	1.8
	505	100.1%			210	99.9%	337	99.9%	44	100.0%	1,096	100.0%

LODGING ACCOMMODATIONS

The 1974 airline skier survey shows slightly different figures. In 1974, lodges were the leading type of accommodation, while in 1978 the trend toward more condominiums has caused them to catch lodges as a place to stay. There was also a large increase in the number staying with friends in 1978, as compared to 1974. See Table 25.

TABLE 25

TYPE OF ACCOMMODATIONS, 1974 AIRLINE SKIER SURVEY

	Number	Percent
Lodge, motel, hotel	460	46.9%
Condominium	407	41.5
Dormitory	22	2.2
Rented a house	17	1.7
Stayed with friends	57	5.8
Camper	3	0.3
Other	15	1.5
	981	99.9%

Accommodation Reservations

Table 26 shows the time frame in which commercial air respondents made lodging reservations. Over 56 percent made their reservations two or more months in advance with 30.0 percent making them four or more months in advance.

TABLE 26

	А	Aspen Va		ail	Winter Park Ste			Copper Steamboat Mountain			Total		
	#	%	#	%	#	%	#	%	#	%	#	%	
4 months or more	169	33.5%	109	30.0%	77	36.6%	78	23.2%	5	11.3%	438	30.0%	
2-3 months	151	29.9	93	25.6	46	21.9	78	23.1	15	34.1	383	26.3	
1 month	66	13.1	51	14.1	17	8.1	43	12.8	20	45.5	187	12.8	
Less than 1 month	93	18.4	68	18.7	27	12.9	51	15.1			249	17.1	
Other		_	17	4.7	4	1.9	8	2.4	1	2.3	30	2.1	
No reservation	-	-	25	6.9	35	16.7	72	21.4	-	-	132	9.1	
No response	26	5.1	_	_	4	1.9	7	2.1	3	6.8	40	2.7	
	505	100.0%	363	100.0%	210	100.0%	337	100.1%	44	100.0%	1,459	100.1%	

ADVANCE LODGING RESERVATIONS

COMMERCIAL AIR SKI VACATION CHARACTERISTICS

Nights Stayed

Table 27 shows the distribution of length of stay. The most frequently occurring value was 7 nights with over 30 percent. A total of 19 percent stayed 6 nights, and over 23 percent stayed between 8 and 14 nights. About 14 percent stayed 4 to 5 nights. Over 92 percent of the respondents stayed 4 or more nights, and 79 percent stayed 6 or more nights. Vail and Aspen recorded the longest length of stays.

TABLE 27	7
----------	---

NIGHTS SPENT AT SKI RESOF	NIGHTS	SPENT	AT	SKI	RESORT
---------------------------	--------	-------	----	-----	--------

					Wi	nter			Co	pper		
	A	spen	V	Vail Park			k Steamboat			ntain	Tot	al
	 #	%	#	%	#	%	#	%	#	%	#	%
None	2	0.4%	5	1.4%	18	8.6%	1	0.3%	-	- %	26	1.8%
1 night	2	0.4	3	0.8	2	1.0	1	0.3	-	-	8	0.6
2 nights	3	0.6	9	2.5	5	2.4	6	1.8	-	-	23	1.6
3 nights	8	1.6	11	3.0	16	7.6	15	4.5	1	2.3	51	3.4
4-5 nights	39	7.7	45	12.4	42	20.0	134	39.7	7	15.9	200	13.7
6 nights	- 93	18.4	63	17.4	50	23.8			. 4	9.1	277	19.0
7 nights	187	37.0	112	30.9	54	25.7	163	48.4	10	22.7	444.5	5 30.5
8-14 nights	139	27.5	95	26.2	.15	7.1					337.8	3 23.2
15-21 nights	17	3.4	15	4.1	2	1.0	14	4.2	22	50.0	48.3	3.3
22 or more nights	11	2.2	4	1.1	4	1.9					~ 33.4	4 3.0
No response	4	0.8	1	0.3	2	1.0	3	0.9	-		10	0.7
	505	100.0%	363	100.1%	210	100.1%	337	100.1%	44	100.0%	1,459	100.8%

V

34

Length of visit figures from the 1974 airline skier survey are shown in Table 28. It is necessary to group these figures into the categories shown in Table 27. It appears the figures are very comparable when one adjusts the 6 night stay into a 7 day length of stay.

TABLE 28

LENGTH OF VISIT, 1974 AIRLINE SKIER SURVEY

	Number	Percent
l day	1	0.1%
2 days	11	1.1
3 days	16	1.6
4 days	32	3.3
5 days	48	4.9
6 days	52	5.3
7 days	555	56.6
3 days	128	13.0
9 days	65	6.6
10 days	18	1.8
ll days	6	0.6
12 days	4	0.4
13 days	10	1.0
14 days	20	2.0
15 days	3	0.3
16 days or more	$\frac{12}{981}$	<u> 1.2</u> 99.8%

Package Plan

Approximately 23.0 percent of the commercial air respondents were on a package plan. The figures ranged from a low of 19.0 percent at Aspen to a high of 31.8 percent at Steamboat, as shown in Table 29. These figures are markedly different from the 1974 survey which recorded 55 percent of the respondents on a package plan. The lack of package plans available in the 1977-78 season compared to the 1973-74 season had a great impact on the results.

TABLE 29

ON A PACKAGE PLAN

	٨	0202	Ţ	ail		nter ark	Sto	amboat	-	per	То	tal
	 #	spen %	#	<u>%</u>	#	<u>%</u>	#	%	#	%	#	%
Yes No No response	96 408 <u>1</u> 505	19.0% 80.8 0.2 100.0%	74 287 <u>2</u> 363	20.4% 79.0 <u>0.6</u> 100.0%	49 157 <u>4</u> 210	23.3% 74.8 <u>1.9</u> 100.0%	107 220 <u>10</u> 337	31.8% 65.3 <u>3.0</u> 100.1%			326 1,072 <u>17</u> 1,415	23.0% 75.8 <u>1.2</u> 100.0%

Ski Vacations

Table 30 shows the ski vacations taken last season by commercial air respondents. The largest group (34.6 percent) took one ski vacation, followed by the group taking two (23.2 percent). A ski vacation has been defined as a stay of four or more nights.

TABLE 30

					Wi	nter			Cor	oper		
	A	spen	V	ail	P	ark	Ste	amboat	Mour	ntain	Tot	al
. <u> </u>	#	%	#	%	#	%	#	%	#	%	#	%
None	11	2.2%	_	- %	8	3.8%	158	46.9%	11	25.0%	188	14.6%
1 vacation	249	49.3	4	2.1	121	57.6	108	32.0	23	52.3	505	39.3
2 vacations	147	29.1	93	49.5	45	21.4	47	13.9	6	13.6	338	26.3
3 vacations	56	11.1	53	28.2	17	8.1	9	2.7	2	4.5	137	10.7
4 vacations	25	5.0			9	4.3	8	2.4	2	4.5	56.0	5 4.4
5 vacations	7	1.4	38	20.2	7	3.3	3	0.9	-	-	26.2	2 2.0
6 or more vacations	10	2.0					3	0.9	_	-	29.2	2 2.3
No response	-	-	2	0.6	3	1.4	1	0.3			6	0.5
	505	100.1%	190	100.6%	210	99.9%	337	100.0%	44	99.9%	1,286	100.1%

NUMBER OF SKI VACATIONS (FOUR NIGHTS OR MORE)

Table 31 shows the number of ski vacations taken by respondents in the 1974 survey; however, the reader should be aware that a different definition of ski vacation was used. In the 1974 study a ski vacation was defined as requiring a minimum of one overnight stay while the 1977-78 studies required a stay of four or more nights.

	Number	Percent
1 vacation	364	37.1%
2 vacations	230	23.4
3 vacations	148	15.1
4 vacations	91	9.3
5-8 vacations	82	8.4
9-11 vacations	12	1.2
12 or more vacations	47	4.8
No response	7	0.7
	981	100.0%

NUMBER OF SKI VACATIONS TAKEN PER YEAR, 1974 AIRLINE SKIER SURVEY

TABLE 31

Brochures

Four of the areas asked respondents if they had seen a brochure before coming, and 42.5 percent of the commercial air respondents indicated they had, as shown in Table 32. This was the highest percentage reported by any of the market segments analyzed.

TABLE 32

BROCHURE	SEEN
----------	------

	Aspen		v	WinterCopperVailParkSteamboatMountain			Total					
	#	%	#	%	#	%	#	%	#	%	#	%
Yes			197	54.3%	78	37.1%	124	36.8%	6	13.6%	405	42.5%
No			165	45.5	127	60.5	123	36.5	38	86.4	453	47.5
No response			1	0.3	5	2.4	90	26.7			96	10.1
			363	100.1%	210	100.0%	337	100.0%	44	100.0%	954	100.1%

Sources of Planning Information

When out-of-state respondents were asked about their sources of planning information, "word of mouth" was the prime source (30.9 percent), as shown in Table 33. Other important sources were travel agents, brochures, airlines and advertisements. The totals in Table 33 add to more than 100 percent because of multiple responses.

TABLE 33

SOURCES OF PLANNING INFORMATION

					Wi	nter			Co	pper		
	A	spen	V	ail	P	ark	Ste	amboat	Mou	ntain	То	tal
	#	%	#	%	#	%	#	%	#	%	#	%
Airline			86	23.7%	11	5.2%	29	8.6%	7	15.9%	133	9.6%
Travel agent			167	46.0	26	12.4	58	17.2	6	13.6	257	18.6
Brochures			43	11.8	19	9.0	71	21.1	2	4.5	135	9.8
Radio/TV ads			3	0.8	_	-			-	-	8	0.6
Magazine ads			26	7.2	6	2.9	10	3.0	1	2.3	38	2.8
Past experience			-	_	90	42.9	_	-	-	-	90	6.5
Word of mouth			163	44.9	85	40.5	156	46.3	23	52.3	427	30.9
Ski show			1	0.3	_	-	11	3.3	1	2.3	13	0.9
Ski club/auto			3	0.8	-	-	33	9.8	1	2.3	37	2.7
Other			108	29.8	22	10.5	77	22.9	10	22.7	217	15.7
None			10	2.8	_	-	-	-	_	_	10	0.7
Not applicable			_	_	11	5.2	_	-	_		11	0.8
Source unknown			_	-			6	1.8	_	-	6	0.4
boules and wit			610 ^a	 168.1% ^a	270 ^a	128.6% ^a	451 ^a	134.0% ^a	a	115.9 ^a	1,382	100.0%

^a Multiple responses.

Travel Arrangements

Table 34 shows the use of various agencies in making travel arrangements of commercial air respondents. The largest group (31.5 percent) reported using a travel agent; the second largest group (20.8 percent) made their reservations direct. The role of the travel agent is very important in dealing with skiers traveling by commercial air. Because of multiple responses, some of the totals in Table 34 do not add to 100 percent.

TABLE 34

					Wi	nter			Co	pper		
	А	spen	V	ail	Р	ark	Ste	amboat	Mountain		То	tal
	#	%	#	%	#	%	#	%	#	%	#	%
Travel agent	125	24.8%	188	51.8%	42	20.0%	93	27.6%	11	25.0%	459	31.5%
Airline	22	4.4	52	14.3	4	1.9	-		8	18.2	86	5.9
A friend	-	_		_	24	11.4	33	9.8	15	34.1	72	4.9
Area association	-	-	27	7.4	14	6.7	11	3.3	5	11.4	57	3.9
Direct	127	25.2	46	12.7	63	30.0	64	19.0	3	6.8	303	20.8
Ski club	5	1.0	11	3.0	11	5.2	41	12.2	2	4.5	70	4.8
None made	-	-	-	-	3	1.4	72	21.4	-	-	75	5.1
Other	133	26.3	69	19.0	7	3.3	6	1.8	2	4.5	217	14.9
Not applicable	_	-	-	_	28	13.3	-	-	_	-	28	1.9
No response	11	2.2	8	2.2	14	6.9	17	5.0	_	-	50	3.4
Aspen reservation a	agencies 82	16.2	-	-	-	-	-	_	-	-	82	5.6
	505	100.1%	401	110.4%	210	100.1%	337	100.1%	46	104.5%	1,499	102.7%

First Ski Vacation

When respondents were asked if this was their first ski vacation at the area, the results were split about 50-50, as shown in Table 35. Copper Mountain and Steamboat attracted over two-thirds of their commercial air visitors for the first time. Aspen and Vail attracted about 40 percent while Winter Park attracted 50 percent.

TABLE 35

FIRST SKI VACATION

					nter			Co				
	A	spen	V	Vail		Park St		Steamboat 1		ntain	Total	
	 #	%	#	%	#	%	#	%	#	%	#	%
Yes	203	40.2%	145	39.9%	105	50.0%	226	67.1%	30	68.2%	709	48.6%
No	302	59.8	216	59.8	104	49.5	111	32.9	14	31.8	747	51.2
No response	505	-	$\frac{1}{362}$	$\frac{0.3}{100.0\%}$	$\frac{1}{210}$	$\frac{0.5}{100.0\%}$	337	-	44	-	$\frac{2}{1,458}$	$\frac{0.1}{99.9\%}$

QUESTIONNAIRES

APPENDIX A

ASPEN SKIER INTERVIEW STUDY, 1977-78 SEASON

Name	of Inter	rviewer		Da	1te
Name	of Mount	ain: 1 Aspen 2 Highla	ands	3 Snowmass	4 Buttermilk
1.	Where an	re you from?			
		City			State
0				•	
2.	What is	the number of people in your	party	6	
	Children	n Teens	Adults	s TOTA	AL
3.	If you a	are from out of state, how dia	d you :	reach Colorado?	
	1	From Colorado	6	Scheduled bus	
	2	Train	7	Chartered plane	
	3	Rental car	8	Private plane	
		Private car	9	Commercial air	
	5	Chartered bus			
4.	If you o	came by air, which airline di	d you	use to get to Co	lorado?
	1	United	8	North Central	
	2	Continental	9	Ozark	
	3	Frontier	10	Delta	
	4	TWA	11	Other (specify)
	5	Western	12	Didn't use comm	ercial airline
	6	Braniff	13	From Colorado	
	7	Texas International			
5.	What ai:	rport did you fly into to get	to Co	lorado?	
,	1	Denver	5	Aspen	
		Grand Junction		Other (specify)
		Colorado Springs	7		
	4	Salt Lake City		From Colorado	
6.	How did	you arrive in the Aspen area	?		
	1	Private car	5	Express bus fro	m Stapleton Airport
		Rental car	6		
	3	Chartered bus	7	Private plane	
	4	Scheduled bus		Aller Martine and	
		A ANDLES TRUE LODGE			
7.	Are you	using a car while you are in	Aspen	?	
	1	Yes, own car	3	Yes, other car	(specify)
	2	Yes, rental car	4	No	

8.	Do you feel a car is needed?	
	1 Did not answer yes 2 Yes	3 No
9.	Is your lodging in:	
	1 Aspen area 2 Snowmass Resort 3 Basalt	4 Glenwood Springs 5 Other (specify)
10.	Who did you contact in making you find out who actually made the re	r lodging reservation? (Probe if necessary, eservation.)
	1 Aspen Reservations, Inc.	7 Airlines
	2 Snowmass Central Reservat	tions 8 Travel Agent
	3 Vilcor	9 Direct to accommodations
	4 Aspen Ski Tours	10 Ski club (specify)
	5 Aspen Condo Reservation 6 Chamber of Commerce	<pre>11 Other (specify tour operator's</pre>
11.	How far in advance of your arriva	al date did you book space in Aspen?
	1 More than 5 months	6 One month
	2 Five months	7 Three weeks
	3 Four months	8 Two weeks 9 One week or under
	4 Three months 5 Two months	9 One week or under
	5 IWO MOLICIIS	
12.	What type of accommodations do ye	ou have on this trip?
	1 Lodge, motel, hotel	5 Staying with friends
	2 Condominium	6 Camper
	3 Dormitory	7 Other (specify)
	4 Renting a house	
13.	Do you like the City of Aspen's	pedestrian mall?
	1 Yes	3 No opinion
	2 No	4 Haven't been in Aspen yet (go to
		question 17)
14.	If yes, should the mall:	
	1 Be expanded	3 No opinion
	2 Remain the same size	4 Not applicable
15.	Do you find parking in the downt	
	1 No problem	4 Not using a car
	2 A little difficult	5 Not applicable
	3 Very difficult	
16.	Does Aspen's maximum of two hour the downtown area?	parking allow you adequate time to shop in
	1 Yes	4 Not using a car
	2 Usually	5 Not applicable
	3 No	

17.	How many ski vacations do you take on t includes a minimum of four nights.)	the	average each year? (A ski vacation
	1 None	5	Four
	2 One	6	Five
	3 Two	7	Over five
	4 Three		
18.	How many total days do you actually spe daily and vacation ski trips.)	end	skiing per year? (Include both
	1 Three or less	5	Fifteen to twenty-one
	2 Four to six	6	
	3 Seven to ten	7	
	4 Eleven to fourteen		
19.	How do you classify yourself as a skier	r?	
	1 Novice	4	High intermediate
	2 Beginner	5	Advanced
	3 Low intermediate	6	Expert
20.	Are you a member of a ski club?		
	1 Yes	2	No
21.	Is this your first ski vacation to Aspe	en?	
	1 Yes	2	No
22.	Did you ski or will you ski at any othe	er (Colorado areas on this trip?
	1 Yes	2	No
	3		
23.	If yes, where?		
24.	How many nights will you spend in the	Asp	en-Snowmass Resort area?
25.	How many days will you take lessons th	is :	ski trip?
	1 None	5	Four
	l None 2 One	6	Five
	3 Two	7	the second s
	4 Three	8	
	4 111126	0	Seven or more
26.	If yes, where?		
27.	Who are you here with? (Show responde category.)	nt	the list, have them indicate which
	1 An organized group, i.e.,	4	Friends
	a ski club	5	
	2 Your family	6	
	3 Your family and friends	7	

28.	How	man	y of your group are non-ski	iers?		
	r B	1 2 3 4 5	Three	8	7 8 9	Nine to ten
29.	Are	you	on a package plan?			
		1	Yes (go on to question 30)	:	2	No (go on to question 33)
30.	If y	es,	please indicate your first	t choice	e	of ski package length.
		2	Three days Five days Six days Seven days	(Other (specify) Doesn't matter Not applicable
31.	Your	se	cond choice:			
		2 3	Three days Five days Six days Seven days	(Other (specify) Doesn't matter Not applicable
32.	What	is	the total cost of your page	ckage or	n	a per person basis?
		1	Excluding transportation		\$	
		2	Including transportation		\$	
33.			indicate how much <u>YOU</u> or so lowing items:	omeone	el	se spent on yourself yesterday for
	(a)	Lo an	dging (non-package plan us d are paying \$42.50 a nigh	er). E t for a	xa cc	mple: If you are a party of four ommodations, the figure is \$10.63.
		\$_		Don't K	no	w
	(b)	Re	estaurant (breakfast, lunch	and di	nn	er, including alcoholic beverages).
		\$		Don't K	no	W
		In	dicate places they ate:			
			Breakfast			
			Lunch			
			Dinner			

34.	Please estimate how much YOU for the following items this items "a" through "g;" if ans it in.)	ski tr	ip. (Interv	viewer: Be	sure to co	ver all
	(a) Groceries \$		(f) Ski	lessons	\$	
	(b) Package liquor \$		(g) Othe	er (rental	car.	
	(c) Clothing \$	2		ysitting, f		
			-	ts, souveni		
	(d) Equipment \$. mea	ical, etc.)	ې	
	(e) Ski rental \$					
35.	How much did you spend on you package plan user only.)	rself	for round ti	rip transpo	rtation? (Non-
	\$		e-mijie e			
36.	How does the Aspen experience	rate '	with other s	ski vacatio	ns you have	taken?
	 Much better Better About the same Not as good as most 		6 Firs	n worse st ski trip e vacatione		spen
37.	Do you plan to return to Aspe	n for	a ski vacati	ion in the	future?	
	1 Yes, again this year 2 Yes, next year 3 Don't know	i Conglis	5 Some 6 No	etime in th		years
38.	Do you feel welcome in Aspen	when u	sing the fol	llowing fac	ilities?	
	Controller - Sec				Generally	
	<u> </u>	lways	<u>Generally</u>	Not	Not	Contact
	(a) On the ski mountain?	1	2	3	4	5
	(b) At the lift ticket	1	0	2	1	F
	window?	1	2	3	4	5
	(c) In the mountain restaurants?	1	2	3	4	5
	(d) At the lodge/condo?	1	2	3	4	5
	(e) In shops?	<u>+</u>	2	3	4	5
		<u> </u>	2	3	4	5
			Z	J	<u></u>	
	(g) At the Chamber of	1	2	2	4	5
	Commerce?	$\frac{1}{1}$	2	33	4	<u>5</u>
	(h) At the Airport?	<u>L</u>	<u>L</u>	<u>ى</u>	4	
	(i) At City Hall/Police	1	0	2	1.	5
	Dept./Sheriff's Office?	1	2	3	4	5
	(j) In taxis?	1	2		the second s	5
	(k) In buses?	1	2	3	4	<u>></u> 5
	(1) In ski school?	1	2	3	4	2

39. What is your opinion of Aspen for each of the points listed below?

	Excellent	Good	Average	Fair	Poor	Not Applicable
		0000			1001	mppiicabie
(a) Lodging accommodations	1	2	3	4	5	6
(b) Lift line lengths	1	2	3	4	5	6
(c) Trail grooming	1	2	3	4	5	6
(d) Variety of trails	1	2	3	4	5	6

40. Have you ever taken a summer vacation in the Aspen area?

1 Yes 2 No

41. Would you consider visiting Aspen in the summer?

1 Yes 2 No

42. If you were going to spend a summer vacation in the Aspen area, how important would the availability of the following facilities, activities and programs be in your decision to come in the summer? Not

				NOE
		Important	Undecided	Important
(a)	Ballet	1	2	3
(b)	Tennis courts	1	2	3
(c)	Golf	1	2	3
(d)	Summer stock theatre	-1	2	3
(e)	Miniature golf	1	2	3
(f)	Pedestrian mall	1	2	3
(g)	Recreational vehicle park	1	2	3
(h)	Professional seminars/conferences	1	2	3
(i)	Jeep tours		2	3
(j)	Trout fishing	1	2	3
(k)	Backpacking trips	1	2	3
(1)	Chairlift rides	1	2	3
(m)	Aspen Music Festival	1	2	3
(n)	Local shuttle bus service	1	2	3
(0)	Horseback riding	1	2	3
(p)	Raft trips	1	2	3
(q)	Arts and crafts	1	2	3
$\frac{1}{(r)}$	Scenic tours	1	2	3
(s)		1	2	3

43. Sex: 1 Male 2 Female

44. Are you married? 1 Yes 2 No

- How many years have you been skiing? 45.
 - Less than one 1
 - 2 One

- 5 Seven to nine
- 6 Ten to twelve 7

- 3 Two to three
- 4 Four to six

Thirteen to fifteen

46. How old are you?

1	Under	12		6	30-	-39	
2	13-17			7	40-	49	
3	18-19			8	50-	-59	
4	20-24			9	60	or	over
5	25-29			10	No	ans	swer

47. What is the annual income of your household?

1	\$0-6,999	6	\$25,000-34,999
2	\$7,000-9,999		\$35,000-49,999
3	\$10,000-14,999	8	\$50,000-74,999
4	\$15,000-19,999	9	\$75,000 and over
5	\$20,000-24,999	10	No answer

48. What is the highest grade you have completed in school?

1	0-8th grade	5	College graduate
2	Some high school	6	Post-graduate work
3	High school graduate	7	Advanced degree
4	Some college	8	No answer

49. What is your occupation? (If in doubt on classification, write in occupation.)

- 1Executive, manager6Secretary, office work2Professional, technical7Crafts, factory, mechanic3Teacher, professor8Homemaker
- 4 Salesman, buyer
- 5 Government, military
- 9 Student
- 10 Other
- 50. Do you have any comments about Aspen-Snowmass area which would help make the area a better place to ski?

51. Which of the following publications do you read on a regular basis? (Check all that apply.)

	Ski		Oui	Denver Magazine
	Skiing		Esquire	Power
	Time	2111	Mademoiselle	Barrons
	National Geographic		Vail Trail	 Aspen Times
	Denver Post		Christian Science Monitor	 New Yorker
	Newsweek		Colorado Magazine	 Glamour
	Playboy		Business Week	 Viva
	Rocky Mountain News		Family Circle	 Your local metro
	Sports Illustrated	-	Reader's Digest	 newspaper
	Penthouse		Texas Monthly	Other (specify)
ſ	U.S. News & World Report		Southern Living	
	People		Sunset	49
	Wall Street Journal		Chicago Magazine	

52.	Have you seen an advertisement on winter in Aspen in the last 8 months?
	1 Yes 2 No
53.	Have you ever seen a ski movie on Aspen?
	1 Yes 2 No
54.	If yes, where?
	1 Television5 Movie theatre2 Ski club meeting6 Airplane3 Ski show7 Other (specify)4 Airport
55.	Would you be willing to participate in similar follow-up questionnaires if they were to be mailed to your home?
	1 Yes (see a) 2 No
	(a) Name
	Address
	City and State
56.	Would you like to receive information on conference facilities in Aspen? 1 Yes 2 No
	If yes, record address:

Thank you for your cooperation.

VAIL SKIER SURVEY

	LOCAL	1977-1978 COLORADO SI	JRVEY
	so a set i so a hor and		NAME
			DATE
		6.	CHAIR TIME
1.	Where do you live?		
1 •	where do you rive:		
	CITY	STATE	ZIP
2.	How did you get here?		
	(1)Private car (2a&b)	(5)	Commercial Airline:
	(2) Rental car (2a&b)		What airline
	(3) Scheduled bus	(6)	Charter airline
	(4) Scheduled bus		What airline
	(5) Charter bus	(7)	Private Plane
	ter en ere erete på bere deberer i	(8)	Train
	2a. How many people, inclu	ding yourse	lf, were in your car?
	the state of the second s		2.6 (a 20 million.)
	TotalAdul (18 and		TeensChildren13 - 17)(12 and under)
	2b. Where is it parked?		
	(1)Golden Peak	(4) _	Private lot
	(2)Transportation	(5)	Not applicable
	Center	(6)	Don't know
	(3)LionsHead	18 C	Charles In Information
3.	How do you classify yoursel	f as a skie	er? a conduct notage to the quality
	(1) Novice (never-ever)	(4)	High Intermediate (parallel)
	(2) Beginner	(5)	Advanced
			Expert (professional)
	(stem christy)	(0)_	
4.	How many total days will yo	u ski durin	ng this <u>entire ski season</u> ?
	(1) None	(6)	6 days
	(2) One day	(7)	7 days
	(3)Two days	(8)	8 to 14 days
	(4) Three days	(9)	15 to 21 days
	(5)4 to 5 days		22 or more
5.	Are you taking ski school]	essons?	
		1.0.1	

(1) ____Yes (5a)

(2) No (5b)

5a. On this scale, how would you rate the quality of instruction?

	<pre>(1)Outstanding (2)Excellent (3)Good (4)Average</pre>	(5)Fair (6)Poor (7)Don't know
5Ъ.	Why?	
	(1)Don't need (2)No time (3)Cost	<pre>(4)Planning to take lessons (5)Other, please specify</pre>

6. What kinds of new ski school programs would you like to see added?

7. With regard to lift lines, how many minutes do you consider an unacceptable wait?

(1)	1 to 5 minutes	(4)16	to	20 minutes
(2)	6 to 10 minutes	(5) 21	to	29 minutes
(3)	11 to 15 minutes	(6)30	or	more

8. With regard to chair lift rides, how many minutes do you consider comfortable?

(1)	3	to	6 minutes	(3)	10	to	12	minutes
(2)	7	to	10 minutes	(4)	13	to	15	minutes
				(5)	Ove	er l	L5 1	ninutes

- 9. How significant a role do you think Vail's Gondola plays in the quality of the total skiing experience?
 - (1) A necessity

 - (2) Plays an important role
 (3) Adds a touch of class
 (4) Has no material signific. Has no material significance
 - (5) Has a negative connotation
- 10. With a view toward preserving a high quality ski experience, what would be your general reaction to a reservation system that limited the number of skiers on Vail Mountain on a given day?

(1) _	Approve	(2)	Disapprove	(10a)	(3)	_Dont' know
10a.	Why?		63	20		

11.	How did you purchase your lift t	icket?	
	(1) Package plan thru lodge	(5)	Season Pass
	(2) Daily ticket at ticket	(6)	Payless Pass
	office	(7)	Group rate
	(3) Multiple day at ticket	(8)	Other, please specify
	office	(0)	
	(4) Colorado Card		
			like in the second s
2.	How many nights will you spend i	n the V	ail area?
	(1) None	(6)	6 nights
	(2) One night	(7)	7 nights
	(3) Two nights	(8)	8 to 14 nights
	(4) Three nights	(9) —	
	(5) 4 to 5 nights	(10) —	22 or more
	(J)4 to J nights	(10)	22 01 more
3.	Where are your accommodations?		
	(1) Vail Village	(5)	West Vail (13a)
	(2) LionsHead	(6)	
	(3) East Vail (13a)	(7)	Other
	$(4) \qquad \text{Sandstone (13a)}$	(8) -	Not applicable
	(4)Sandstone (15a)	(0)	NOL applicable
	13a. Do you find a need for a p	16-	Trives
	(1)Yes	(2)	No
4.	How many people are living/stay:	ing in y	your lodging accommodations?
	(14a)		Patr
	14a. How many do not ski?		
5.	In the past year, have you seen (check all that apply)	or hear	d any of the following?
	(1) Vail brochure	(4)	Direct mailing
	(2) Magazine advertising	(5)	Ski the Rockies advertising
	(3) Radio (not KVMT, local	(6)	Other
	radio station)	(7)	None
6.	On this scale, what is your opin brochures?		
	(1) Outstand 11	(1)	
	(1)Outstanding	(4)	Average
	(2)Excellent	(5)	Fair
	(3)Good	(6)	Poor
		(7)	Haven't seen

17. How would you describe Vail's image?

18. Have you ever skied Vail in April (do not ask during April)

(1) Yes (18a) (2) No (18b)

18a. What was your opinion of the skiing experience?

(1) Outstanding	(4)	Average
(2) Excellent	(5)	Fair
(3) Good	(6)	Poor

18b. What do you think the skiing would be like?

(1)	Outstanding	(4)	Average
(2)	Excellent	(5)	Fair
(3)	Good	(6)	Poor

19. Have you ever skied Vail in early December? (Dec. 1-17 - Do not ask during November and December)

(1) Yes (19a) (2) No (19b)

19a. What was your opinion of the skiing experience?

(1)	Outstanding	(4)	Average
(2)	Excellent	(5)	Fair
(3)	Good	(6)	Poor

19b. What do you think the skiing would be like?

(1)	Outstanding	(4)	Average
(2)	Excellent	(5)	Fair
(3)	Good	(6)	Poor

20. Would you be willing to park a mile or so from a ski resort and use a free shuttle bus? (Day skier only)

(1) Yes

(2) No

(2) _____Female

21. Sex:

(1) Male

22. Are you married?

(1) Yes (2) No

23. How old are you?

(1)	12	or	under	(5)	25	to	29
(2)	13	to	17	(6)	30	to	39
(3)	18	to	19	(7)	40	to	49
(4)	20	to	24	(8)	50	to	59
				(9)	60	or	over

24. What is the annual income of your household?

	<pre>(1)\$0 to 6,999 (2)7,000 to 8,999 (3)9,000 to 14,999 (4)15,000 to 19,999 (5)20,000 to 24,999</pre>	(6) (7) (8) (9) (10) (11)	25,000 to 29,999 30,000 to 39,999 40,000 to 49,999 50,000 to 74,999 75,000 and over No answer
25.	What is the highest grade of so	chooling	you've completed?
	<pre>(1)0 to 8th (2)Some high school (3)High school graduate</pre>	(4) (5) (6) (7)	Some college College graduate Post graduate work Advanced degree
26.	What is your occupation?		
	<pre>(1)Executive Manager (2)Professional/Technical (3)Teacher/Professor (4)Salesman/Buyer (5)Government/Military</pre>	(6) (7) (8) (9) (10) (11)	Office work/Secretary Crafts/Factory/Mechanic Homemaker Student Service organization Other, please specify
27.	Do you have any suggestions as	to how t	ve can improve:
		ang genera Ion	Teurs Children
28.	Would you be willing to partic	ipate in	this type of questionnaire by mail?
	(1) <u>Y</u> es	(2) _	No
	NAME	165	
	ADDRESS	0 h 200	
	CITY/STATE/ZIP		

		KIER SURVEY	(
		-STATE SURV	EV	
	001-01-	STALE SURV.	NAME	
			MAPIL	
			DATE	
			CHAIR	TIME
Where do you live	≥?			
CITY		STATE		ZIP
How did you reach	Colorado?			
now and you react	I COIDIAUD:			
(1) Private o	car	(5)	Commercial airl	ine
(2) Rental ca			What airline	
(3) Scheduled		(6)	Charter airline	
(4) Charter h	bus		What airline	
the second s		(7)	Private plane	
		(8)	Train	
			ph. In the might have	
Once within Color	rado, how did	you get in	to the Vail area	?
	i sona da loci			
	car (3a&b)	(5)	Express bus fro	m Stapleton
	ar (3a&b)	(6)		
(3) Schedule		Contraction of the local division of the loc	Private plane	
(4)Charter 1	bus	(8)	Other, please s	pecify
3a. How many p	eople, includ	ing voursel	f, were in your	car?
Sur now many p	coprey increa	ing jourdes	il, weie in your	cur .
Total	Adu	lts	Teens	Children
	(18 and	over)	(13 - 17)	(12 and under)
3b. Where is i	t parked?			
and a second strategy of	_		and and a second s	
(1)Go	lden Peak	(4)	Private lot	
(1)Go (2)Tr	lden Peak ansportation	(5)	Not applicable	
(1)Go (2)Tr Ce	lden Peak ansportation nter			
(1)Go (2)Tr Ce	lden Peak ansportation	(5)	Not applicable	
(1)Go (2)Tr Ce (3)L1	lden Peak ansportation nter onsHead	(5) (6)	Not applicable Don't know	
(1)Go (2)Tr Ce	lden Peak ansportation nter onsHead	(5) (6)	Not applicable Don't know	
(1)Go (2)Tr Cer (3)1 . How do you class	lden Peak ansportation nter onsHead	(5) (6) as a skier?	Not applicable Don't know	
(1)Go (2)Tr Cer (3)1 . How do you class (1)Novice (lden Peak ansportation nter onsHead dify yourself never-ever)	(5) (6) as a skier (4)	Not applicable Don't know High intermedia	
(1)Go (2)Tr Ce (3)Li . How do you class (1)Novice ((2)Beginner	lden Peak ansportation nter onsHead ify yourself never-ever)	(5) (6) as a skier (4) (5)	Not applicable Don't know High intermedia Advanced	
(1)Go (2)Tr Cer (3)Li . How do you class (1)Novice ((2)Beginner (3)Low inte	lden Peak ansportation nter onsHead ify yourself never-ever) ermediate	(5) (6) as a skier (4)	Not applicable Don't know High intermedia	
(1)Go (2)Tr Ce (3)Li . How do you class (1)Novice ((2)Beginner	lden Peak ansportation nter onsHead ify yourself never-ever) ermediate	(5) (6) as a skier (4) (5)	Not applicable Don't know High intermedia Advanced	
(1)Go (2)Tr Cer (3)1 . How do you class (1)Novice ((2)Beginner (3)Low inte (stem ch	lden Peak ansportation nter onsHead ify yourself never-ever) ermediate risty)	(5) (6) as a skier? (4) (5) (6)	Not applicable Don't know High intermedia Advanced Expert	te (parallel)
(1)Go (2)Tr Cer (3)Li How do you class (1)Novice ((2)Beginner (3)Low inte	lden Peak ansportation inter onsHead dify yourself never-ever) ermediate risty) TEW QUESTIONS	(5) (6) as a skier (4) (5) (6) ABOUT YOUR	Not applicable Don't know High intermedia Advanced Expert SKI VACATION. F	te (parallel) LEASE KEEP IN MIND

 (1) None
 (4) 3

 (2) 1
 (5) 4 to 6

 (3) 2
 (6) 7 or more

6. Do you generally take more than one ski vacation per year? (1) Yes (6a) (2) No 6a. How many? (1) ____One (2) ____Two (3) ____Three
(4) ___Four or more 7. Do you plan to take a ski vacation to Vail next season? (1) Yes (2) No (3) Undecided 8. How many nights will you stay in Vail this trip?

 (1) ____None
 (6) ____6 nights

 (2) ____One night
 (7) ____7 nights

 (3) ____Two nights
 (8) ____8 to 14 nights

 (4) ____Three nights
 (9) ____15 to 21 nights

 (5) ____4 to 5 nights
 (10) ____22 or more nights

 How many days will you ski Vail this trip? 9.

 (1)
 None
 (6)
 6 days

 (2)
 One day
 (7)
 7 days

 (3)
 Two days
 (8)
 8 to 14 days

 (4)
 Three days
 (9)
 15 to 21 days

 (5)
 4 to 5 days
 (10)
 22 or more days

 (1) ____None (6) ____6 days How many total days will you ski during this entire ski season? 10.

 (1)
 None
 (6)
 6 days

 (2)
 One day
 (7)
 7 days

 (3)
 Two days
 (8)
 8 to 14 days

 (4)
 Three days
 (9)
 15 to 21 days

 (5)
 4 to 5 days
 (10)
 22 or more days

 (2) _____One day (3) _____Two days (4) ____Three days 11. Are you taking ski school lessons this trip? (1) Yes (11a) (2) No (11b) lla. On this scale, please rate the quality of instruction. (1) Outstanding (5) Fair (6) Poor (7) Don't know (2) Excellent (3) _____Good (4) _____Average 11b. Why? (1) ______ Don't need (2) ______ No time (3) ______ Cost (4) Planning to take lessons
 (5) Other, please specify 57 12. What kinds of new ski school programs would you like to see added?

13. With regard to lift lines, realistically, how many minutes do you consider an unacceptable wait?

 (1) _____1 to 5 minutes
 (4) ____16 to 20 minutes

 (2) ____6 to 10 minutes
 (5) ____21 to 29 minutes

 (3) ____11 to 15 minutes
 (6) ____30 or more

 14. With regard to chair lift rides, how many minutes do you consider comfortable? (1) ____3 to 6 minutes (4) ____13 to 15 minutes (5) ____0ver 15 minutes (2) ____7 to 10 minutes (3) ____10 to 12 minutes 15. How significant a role do you think Vail's Gondola plays in the quality of the total skiing experience? (1) ____A necessity (2) Plays an important role
(3) Adds a touch of class (4) Has no material significance
(5) Has a negative connotation With a view toward preserving a high quality ski experience, what would be 16. your general reaction to an advanced reservation system that limited the number of skiers on Vail Mountain on a given day? (1) ____Approve (2) ____Disapprove (16a) (3) ____Undecided 16a. Why? 17. During this particular trip, what vacationing needs or requirements have not been fulfilled? 18. In planning your trip to Vail, what sources of information did you draw upon? (check all that apply) (1) Airline (7) _ Ski shows (2) ____Travel agent
(3) ___Vail brochure (8) _ Automobile club Directories/guide books (9) (4) Magazine or newspaper (10) Radio/TV article (11) Other, please specify (5) Word of mouth (12) None

How far in advance did yo	ou do your general planning for this trip?
(1) More than 12 mont	ths (5) 2 to 3 months
(2) 7 to 12 months	(6) 1 month
$(3) \qquad 6 \text{ months}$	(7) Less than one month
(4) 4 to 5 months	(8) Don't know
	(9) Not applicable
	(9)NOU applicable
	ally make your reservation for this particular
trip?	
(1) More than 4 month	hs (6) 3 weeks
(2) 4 months	(7) 2 weeks
$\begin{array}{c} (4) \\ \underline{} 2 \\ \underline{} 1 \\ \phantom{0$	(9)Other
(5) <u>1</u> month	(10) Don't know
	(11)No reservation
How did you purchase you	r lift ticket?
(1) Package plan thr	u lodge (5) Season Pass
(2) Daily ticket at	
office	
	(7)Group rate
(3)Multiple day at	ticket (8)Other, please specify
office	
(4)Colorado card	
Through whom did you mak	e your transportation and lodging arrangements?
(1) Travel agent	(5) Ski club
(2) Airline	(6) Organized group
(4)Direct with lodg	
	(9) Not applicable
How many people are stay	ing in your lodging accommodations?
23a. How many are not s	kiing?
	nt in selecting your accommodations?
(check only one)	
(1) Front desk servi	
(2) Location	(7) <u>In-house restaurant/bar</u>
(3) <u>Heated pools</u>	(8)Other, please specify
(4) Size of room	
(5) <u>Cost</u>	(9) None

	In the past year, have you seen (check all that apply)	or heard any of the following?
	 Vail brochure Magazine advertising Radio advertising (not KVMT, local radio) 	 (4)Direct mailing (5)Ski the Rockies advertising (6)Other, please specify
		(7)None
	On this scale, what is your opi	nion of Vail's advertising and its brochures
	(1)Outstanding	(5)Fair
	(2) Excellent	(6) Poor
	(3) Good	(7) Haven't seen
	(4) Average	
	2.5 [] it is contain	(m)1 (ca
	How would you describe Vail's i	mage?
	Have you ever skied Vail in Apr	il? (Do not ask during April)
	(1)Yes (28a)	(2) <u>No</u> (28b)
	28a. What was your opinion of	the skiing experience?
	(1) Outstanding	(4) Average
	(2) Excellent	(5) Fair
	(3)Good	(6) <u>Poor</u>
	28b. What do you think the sl	ting would be like? (snow conditions)
	(1) Outstanding	(4) Average
	(2) Excellent	(5) Fair
	(3) Good	(6) Poor woolse to weath
	(0)0000	
•	Have you ever skied Vail in eas during November or December)	rly December? (Dec. 1-17 - Do not ask
	(1) <u>Yes</u> (29a)	(2) <u>No</u> (29b)
	29a. What was your opinion of	f the skiing experience?
	(1) Outstanding	(4) Average
	(2) Excellent	(5) Fair
	(3)Good	(6)Poor
	29b. What do you think the s	kiing would be like? (snow conditions)
		(4)Average (5) Fair

30. Would you be willing to park a mile or so from a ski resort and use a free shuttle bus? (Day skier only)

	(1) <u>Yes</u>	(2) <u>No</u>
31.	Sex:	
	(1)Male	(2)Female
32.	Are you married?	
	(1)Yes	(2) <u>No</u>
33.	How old are you?	X
	(1)12 or under (2)13 to 17 (3)18 to 19 (4)20 to 24 (5)25 to 29	(6)30 to 39 (7)40 to 49 (8)50 to 59 (9)60 or over
34.	What is the annual income of you	r household?
	<pre>(1)\$0 to 6,999 (2)7,000 to 8,999 (3)9,000 to 14,999 (4)15,000 to 19,999 (5)20,000 to 24,999</pre>	<pre>(6)25,000 to 29,999 (7)30,000 to 39,999 (8)40,000 to 49,999 (9)50,000 to 74,999 (10)75,000 and over (11)No answer</pre>
35.	What is the highest grade of sch	nooling you've completed?
	<pre>(1)0 to 8th (2)Some high school (3)High school graduate</pre>	 (4)Some college (5)College graduate (6)Post graduate work (7)Advanced degree
36.	What is your occupation?	
	<pre>(1)Executive/Manager (2) Professional/Technical (3)Teacher/Professor (4)Salesman/Buyer (5)Government/Military</pre>	<pre>(6)Office work/Secretary (7)Crafts/Factory/Mechanic (8)Homemaker (9)Student (10)Service Organization (11)Other, please specify</pre>
37.	Do you have any suggestions as	to how we can improve:
	37a. The mountain operation	р — <i>в</i>

61

37b. Village facilities and services _____

38. Would you be willing to participate in this type of questionnaire by mail?

(1) <u>Yes</u>	(2) <u>No</u>	
NAME	2 R.	
ADDRESS		
CITY/STATE/ZIP	n na a la raciation a la caracteria.	1.2000

STEAMBOAT SKIER SURVEY 1977-1978

INTE	RVIEWER:	Yann			NO .
LIFT					DATE:
*1.	Where are you from?				
1	mere are jou rom				
	CITY	STATE			ZIP
*2.	Sex: (1)	Male		(2)	Female
~ 2 .	Sex: (1)	nale		(2)	I Chiart
*3.	Are you married?	(1) Yes	(2) No		
				(5)	
*4.	-	Steamboat Season Pass			Six day ticket
		Steamboat Club Card			Half day ticket Daily ticket
		Three-day ticket			Complimentary ticket
	(4)	Five-day ticket		(0)	complimentary licket
*5.	How many years have	you been skiing?			
				(5)	Four voorg
		Less than one year			Four years Five - eight years
		One year Two years			Over eight years
		Three years			Don't know
	(4)	Inice years		(0)	
*6.	How do you classify	yourself as a skier?			
	(1) Beginner	(2) Intermedi	ate	(3	3) Advanced
*7.	How did you learn t				
	(1)	Self-taught			Ski club lessons
	(2)	Ski school		(4)	Friends
+0	In comparel which a	long condition do you	nrofor?		
*8.		lope condition do you	prerer:		
		Groomed slopes			Ungroomed powder
		Small moguls			Glade areas
	(3)	Well-developed moguls	5	(6)	Don't know
*9.	How many ski areas	did you ski last seaso	on?		
		None		(5)	Four areas
		One area			Five areas
		Two areas			Six or more areas
		Three areas			Don't know
		Intee diedo		(0)	2011 0 1111 1
*10.	Last year, how many	days did you ski?		_	
*11.	Last year, how many	ski vacations of four	r or more	nigl	nts did you take?
	(1)	None (4) Thre	ee	(7)	Six or more
		One (5) Four	c	(8)	Don't know
	(3)	Two (6) Five	2		
410	De men energie				
*12.	Do you cross countr	y SKI:			

(1) Frequently (2) Occasionally (3) Never tried

STEAM	BOAT SKIER SURVE	Y - 1977-78			1		
*13.	Have you, or do you plan to use the Steamboat Ski Touring Center?						
		(1) Yes		(3) Don't	Know		
		(2) No		(4) Didn'	t know about	Touring	
		and the period of the sec		Cente	er		
*14.	Who are you here	e with?					
		(1) Self (3) Friends	(5) Ski o	lub/other		
) Family/	(6) Other			
		(2) 100021) (1	friends				
*15.	How many persons	s are in your group	?				
	Childre						
		Teenagers					
			Adults		Total		
				<u></u>			
*16.		t any of the follow	ing areas are	e below your	expectations	and are in	
	need of improve			Gergene et al.			
	YES (Improve					- C., 1114.	
	1		1		se area parkin	g facilities	
	2	2	2	• •	cket sellers i School		
	3	3	4	• • • •	ft Operators		
	5	5	5		i Patrol		
	6	6	6		ft Line Length		
	7	7	7		ail signs	of all iteran-	
	8	8	8		ail Grooming		
	9	9	9		underhead Rest	aurant	
	10	10	10	• •	commodation Qu		
	11	11	11		uttle Bus Serv		
	12	12	12	(12) Are	ea Restaurants		
	13	13	13	(13) Ar	ea Entertainme	ent	
	14	14	14	(14) Lo	cal friendline	ss	
*1/.	Did you utiliz	e the free bus serv (1) Regularly	lce				
			(3) No(4) Didn't know about bus				
10 Co.		(2) Occasionally	la ca sera sera	(4) Didn	't know about	bus	
18.	On this trip, h	ow many nights will	you stay?				
	(1) Her	e only for day	(5) Four ni	ghts	(9) Fifteen (or more nights	
	(2) One	-	(6) Five-si		(10) Don't kno	_	
	(3) Two nights (7) Seven-eight nights						
	(4) Three nights (8) Nine-fourteen nights						
19.	Where are you s						
	(1) Fri	onda	(4) Ronted	Condominium	(7) Camper		
	(2) Lod	ge/motel in town ge/motel at mtn	(5) Own con (6) Rented	dominium		accommodations ow	
20.		id you make your reservations?					
			(4) Ski Clu	h/Group	(7) Don't has	ve reservations	
	(1) Fri (2) Tra	end vel agent		to accommo.	(8) Don't kno		
		amboat Chamber	(6) Other		(o) bon c and		
		sort Assn.	(o) other			64	
	Re						

STE	CAMBOAT SKIER SURVEY - 1977-78						
21.	Did you purchase a package plan to	cover transportation, loc	lging and lifts?				
	(1) Yes	(2) No	(3) Don't know				
22.	How far in advance did you make acco	ommodation reservations f	for this trip?				
	(2) 7-12 months	 (4) 4-5 months (5) 2-3 months (6) 1 month 	(7) Less than 1 month(8) Don't know(9) Don't have reservation				
23.	What sources of information did you use in planning your ski trip to Steamboat?						
	(2) Media describing STBT(3) Steamboat movie	(5) Airline(6) Travel Agent(7) Ski Club(8) Advice of a friend	(9) None of the above (10) Don't know				
24.	Did any of the following media stro	ngly influence you to sk:	i Steamboat?				
		(4) Airline Magazine(5) Newspaper Articles(6) Other publications	<pre>(7) T.V. (8) None of the above</pre>				
25.	Have you seen our Steamboat movie, "Steamboat, There's Only One"?						
	(1) Yes	(2) No					
26.	Were any of the following important boat?	considerations in your	decision to ski at Steam-				
	YES NO	DON'T KNOW					
	$\begin{array}{ccc} 1 & 1\\ 2 & 2\end{array}$	1 (1) Snow c 2 (2) Ski te					
	the press of 3 regression are seen 3 million to the	3 (3) Wester	n atmosphere				
	4 4 5 5		l atmosphere endation of friends				
	6 6 7 7	6 (6) Experi	ence on previous visit bility of reservations				
07	T. the sector being the sector of the sector						
27.	(C) but her iffer which he		(4) Shi the Peckice Ad				
	(1) Steamboat Winter Brochure(4) Ski the Rockies Ad(2) Steamboat Magazine Ad(5) Steamboat Summer Ad(3) Steamboat Direct Mailing(5) Steamboat Summer Ad						
28.	Did Steamboat's guaranteed refund p to choose Steamboat rather than ano		now conditions cause you				
	(1) Yes (2) No	<pre>(3) Don't know (4) Didn't know of p</pre>	rogram				
29.	How did you get from your home to D	enver (or another gatewa	y city)?				
	(1) Commercial airline(2) Private auto(3) Bus	(4) Chartered bus(5) Chartered plane(6) Private plane	(7) Rental Car (8) Don't know				
30.	If you flew, which airline did you	use?	65				
	<pre>(1) United (2) Trans World (3) Continental (4) Braniff</pre>	 (5) Frontier (6) Western (7) Ozark (8) North Central 	 (9) Texas International (10) Hughes Airwest/Fronties (11) Delta (12) Didn't fly 				

STE	AMBOAT SKIER SURVEY - 1977-78		
31.	How did you get to Steamboat?		
	(1) Private Car(2) Rental Car(3) Bus	(4) Chartered bus(5) Rocky Mtn. Airways(6) Frontier Airlines	(7) Chartered plane (8) Private Plane (9) Don't Know
32.	How would you describe making yo	ur travel arrangements to St	ceamboat?
	(1) No problems(2) Slight problems	<pre>(3) Serious problems (4) Don't know</pre>	
33.	Last year, how many times did yo	u visit Colorado to ski?	
	<pre>(1) None (2) Once (3) Twice</pre>	(4) Three times(5) Four times(6) Five times	(7) 6 - 10 times (8) Over ten times
34.	How many previous times have you	been to Steamboat to ski?	
	 Never Once Twice 	(4) Three times(5) Four times(6) Five times	(7) 6 - 10 times (8) Over ten times (9) Don't know
35.	Are there any members of your pa small children)?	rty who do not and will not	ski this trip (excluding
	(1) No (2) One Person	<pre>(3) Two or more people (4) Don't know</pre>	
36.	Do you feel a car is necessary d	uring your stay in Steamboa	t?
	(1) Yes	(2) No	(3) Don't know
37.	Are you interested in returning	to Steamboat for a summer v	isit?
	(1) Yes	(2) No	(3) Don't know
38.	How does this ski vacation rate or at other areas?	in overall quality with oth	ers you have taken here
	 Much better Better than most 	(3) Same (4) Not as good	(5) Much worse
39.	Does this trip meet with your ex	pectations prior to coming	here?
	(1) Yes	(2) No	(3) Don't know
40.	Will you return to Steamboat to	ski?	
	(1) Yes	(2) No	(3) Don't know
*41	. What is the highest grade you ha	ve completed?	
	(1) 0 - 8th	(4) Some college	(7) Graduate degree

(2) Some high school(5) College graduate(8) No response(3) High school grad(6) Post-graduate

STEAMBOAT SKIER SURVEY - 1977-78

*42. Which of the following best describes your family incomes?

(1) \$0 - 6,000	(5) \$20,000 - 24,000	(9) Over \$75,000
(2) \$7,000 - 9,000	(6) \$25,000 - 34,000	(10) No answer
(3) \$10,000 - 14,000	(7) \$35,000 - 49,000	
(4) \$15,000 - 19,000	(8) \$50,000 - 74,000	

*43. Which of the following best describes your occupation?

	Executive/manager Professional/		Secretary/office Mechanical/crafts	(9) Teacher (10) Other
	technical			
(3)	Sales	(7)	Homemaker	
(4)	Military/gov't	(8)	Student	

*44. In which of the following age brackets do you belong?

final a section de state

(1)	12 and under	(5)	25-29
(2)	13-17	(6)	30-39
(3)	18-19	(7)	40-49
(4)	20-24	(8)	50-60
(1)	20 24	(0)	50 00

(6) 60 and over

(10) No answer

WINTER PARK SKIER LIFT INTERVIEW STUDY - 1977-78

	1	Winter Park	2 Mary J	ane	Time
When	re a	re you from?			
		City	in .		State
What	t is	the number of peopl	e in your par	ty?	
Chi	ldre	n (12 and under)		Те	ens (13-17)
Adu	lts	(18 and over)	n settingen	TO	TAL
If ·	you	are from out of stat	e, how did yo	ou reach	Colorado?
		From Colorado			Rental car
	2	Commercial air		7	Scheduled bus
		Chartered plane		8	Chartered bus
		Private plane		9	Train
	5	Private car			
If	you	came by air, which a	irline did y	ou use t	o get to Colorado?
	1	United		9	North Central
	2	Continental		10	Ozark
	3	Frontier		11	Air Midwest
	4	TWA		11 12	Other
	5	Western		13	Didn't use a commercial
	6	Braniff			airline
	7	Texas International	L	14	From Colorado
		Delta			
	8	Deita .			
How	dia	l you arrive at the l	Minter Park a	rea?	e.per years " (though both
How	dia 1	l you arrive at the Private car	Vinter Park a	6	Commercial air
How	dia 1	l you arrive at the l	Winter Park a	6	Private air
How	dia 1	l you arrive at the Private car Rental car Scheduled bus	√inter Park a	6	
How	7 dia 1 2 3 4	l you arrive at the Private car Rental car Scheduled bus Chartered bus		6 7 8	Private air
How	7 dia 1 2 3 4	l you arrive at the Private car Rental car Scheduled bus		6 7 8	Private air
	7 dia 1 2 3 4 5	you arrive at the Private car Rental car Scheduled bus Chartered bus Limo service from the	Stapleton Air	6 7 8 port nter Par	Private air Other :k?
	7 dia 1 2 3 4 5	you arrive at the Private car Rental car Scheduled bus Chartered bus Limo service from the using a car while Yes, own car	Stapleton Air	6 7 8 port nter Par 3	Private air Other k? No
	7 dia 1 2 3 4 5	you arrive at the Private car Rental car Scheduled bus Chartered bus Limo service from using a car while Yes, own car	Stapleton Air	6 7 8 port nter Par	Private air Other ck? No
Are	7 dia 1 2 3 4 5 2 900 1 2	you arrive at the Private car Rental car Scheduled bus Chartered bus Limo service from the using a car while Yes, own car	Stapleton Air you are in Wi	6 7 8 port nter Par 3	Private air Other k? No

8.	Did you use the Winter Park Resort Assoc 1 Yes (if yes, go to 8a)	ciation shuttle bus during your stay? 2 No (if no, go to 8b)	
	 8a. If yes, do you believe the price ch 1 Too high 2 A fair price 	harged (50¢) was: 3 Too low	
	8b. If no, were you aware the Winter Pa bus service?	ark Resort Association had a shuttle	
	1 Yes	2 No	
9.	Are you aware that local taxi-limousine 1 Yes	service is available in Winter Park? 2 No	ŀ
10.	What form of transportation do you prefe Winter Park area?	er to use in traveling around the	
	1 Car	4 No preference	
	2 Scheduled shuttle bus service	5 No opinion	
	3 On call taxi or limo service		
11.	How do you classify yourself as a skier	?	
	1 Novice (never-ever)	4 High intermediate (parallel)	l.
	2 Beginner	5 Advanced	
	3 Low intermediate (stem christy)	6 Expert	
	- Marine 193493 (Marine 1447) 14976	* Second Se	
12.	How many years have you been skiing?		
	1 One year or less	5 Five to eight years	
	2 Two years	6 Nine to twelve years	
	3 Three years	7 Over twelve years	
	4 Four years		
13.	How, many ski vacations have you taken or tion includes a minimum of four nights.) B Dug work or under	•
	1 One	5 Five to eight	
	2 Two	6 Nine to twelve	
	3 Three	7 Over twelve	
	4 Four		
14.	How many total days do you actually spe daily and vacation ski trips.)	end skiing per year? (Include both	
	1 Three or less	5 Eleven to fourteen	
	2 Four to five	6 Fifteen to twenty-one	
	3 Six to seven	7 Twenty-two to twenty-eight	
	4 Eight to ten	8 Twenty-nine or more	
	the second second second states that the second		
15.	Is this your first ski trip to Winter P	Park?	
	1 Yes	2 No	
16.	Where did you buy your lift ticket?		
TO.	1 Winter Park	2 Mary Jane	
	T WINCEL LAIK	2 Haly Jane	
17.	Where will you ski today?		
т/.	1 Winter Park	3 Both areas	
	2 Mary Jane	J Dorn aread	
	a nary dance		

18.		ill you buy lunch today?		
	1		6	No lunch
	2		7	
		Mary Jane	. 8	
		Lunch enroute	9	Not applicable
	5	Brown bag		
9	How man	y nights will you spend in the Winte	r Pa	rk area?
	1		6	Six
	2	One	7	Seven
	3	Two	8	Eight to fourteen
	4	Three	9	
	5	Four to five	0	and the second
0.	How man	y days will you ski Winter Park this	tri	n?
••	1		6	-
	2	One	7	Seven
	3		8	
	4		9	The second
	5	Four to five	0	Twenty-two or more
	J		U	iwenty-two or more
1.	When di	d you make your vacation plans for t	his	trip?
	11.1	More than twelve months ago	6	October 1977
	2	January/February/March 1977	7	November 1977
	3	April/May/June/July 1977	8	December 1977
	4	August 1977	9	In 1978
	5	September 1977	0	No Answer
22.	When di	d you make your lodging reservations	for	this trip to the Winter Park
	area?	Presenter of this Lives		•
	1	Over four months ago	6	Three weeks
	2		7	Two weeks
	3		8	One week or under
	4		9	
	5	One month	0	Not applicable
3.	What to	ype of accommodations do you have on	thic	trip? (MAP = Modified
		an Plan; EP = European Plan.)	LIIIS	tip. (mi - ibdilled
	1		6	Staying with friends
	2	Lodge without meals (EP) go to 23b		Camper - go to 23b
		Condominium - go to 23b	8	
		Motel - go to 23b	9	
	5		ň.,	bei epolicable
	23a Ti	f MAP, how do you rate the food?		
	2Ja. 1	Excellent Satisfact	ory	Needs Improvement
	hed you	see a brochurs before mouses your o		No. 10 Comparison Agriculty Conversion
		f not staying in a lodge with meals,	why	not?
	1	the second second second second second second second		Prefer to cook
		Wish more freedom in dining		Room configuration
	3	Prefer the variety of dining in different restaurants	6	Other (please specify)
24.	Name o	f property		<u>a - 6</u>

25.	What typ	e of accommodations would you like	to h	ave on your next trip?
		Lodge with meals (MAP)	6	
	2	Lodge without meals (EP)	7	
		Condominium	8	Other
		Motel	9	Not applicable
		Renting a house	- í	not applicable
	5	Rentiling a nouse		
.6.	Tf atout	na avamiaht is vour laising int		
.0.		ng overnight, is your lodging in:		
		Winter Park		Grand Lake
		Hideaway Park	7	YMCA
		Fraser	8	
		Tabernash	9	Not staying in accommodations
	5	Granby	0	Not staying overnight
7.	Are you	on a pre-paid vacation package pla	n?	
		Yes (if Yes, answer 27a)	2	No
	100 Th	100 (11 100) unover =/u/	-	10
	27a Do	you find the Winter Park vacation	wouch	or a conventioned?
		-		
		Yes	3	Undecided
	2	No		
28.		you contact in making your lodging		ervation? (Probe if necessary,
	find out	: who actually made the reservation	-	
	1	Travel agent	7	Ski organizer
	2	Airline	8	No advance arrangements
	3	Winter Park Resort Association	9	Other, please specify
	4	Direct to accommodations		2
	5	Ski club or organization	0	Not applicable
		Friends or relatives		
	31,8445-1	the thirth is how articles the		
.9.	Who tool	k care of making the reservations	for vo	ur party?
				Group leader
			6	
	2	Husband	0	Other, please specify
		Wife		<u></u>
	4	Friend		
				VederLifes
30.		ning your trip to Winter Park, what	t soul	rces of information did you
	draw up	on? (Check all that apply.)		irk, hand you partly forket
	1	Airline	7	Past experience
	2	Travel agent	8	Ski show
		Winter Park brochure	9	Other, please specify
	4	Radio advertisements		
	5	Magazine advertisements	0	Not applicable
			0	Not appricable
	6	Word of mouth		
31.		see a brochure before making your		
	1	Yes (if Yes, answer 31a)	2	No
				Hot as given as the
	31a. If	yes, did it help influence your d	ecisi	on to come?
	1			No

32. What is your opinion of each of Winter Park's service personnel listed below?

	Excellent	Satisfactory	Need Improvement	No Contact
Ticket sellers	1	2	3	4
Lift operators	1	2	3	4
Ski instructors	1	2	3	4
Ski patrol	1	2	3	4
Hostesses	1	2	3	4
Ski area cafeteria workers	1	2	3	4
Gas station attendants	1	2	3	4
Waitresses and waiters	1	2	3	4
Shop personnel	1 .	2	3	4
Lodge personnel	1	2	3	4
Lodge and condo bus drivers	1	2	3	4
Winter Park shuttle bus drivers	1	2	3	4

33. What is your opinion of Winter Park for each of the points listed below?

		Need	No
Excellent	Satisfactory	Improvement	Contact
1	2	3	4
1	2	3	4
1	2	3	4
1	2	3	4
1	2	3	4
1	2	3	4
1	2	3	4
1	2	3	4
1	2	3	4
1	2	3	4
	Excellent	Excellent Satisfactory 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2 1 2	Excellent Satisfactory Improvement 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3 1 2 3

- 34. If indoor ice skating were available at Winter Park, would you participate?
- 35. If indoor tennis were available at Winter Park, would you participate? Yes _____No _____Undecided
- 36. If summer tennis clinics were held at Winter Park, would you participate? ____Yes ____No _____Undecided
- 37. How do Winter Park prices compare to other areas?1Much higher2Higher3Same6Don't know
- 38. How does the Winter Park experience rate with other ski trips/vacations you have taken?

1	Much better		4	Not as good as a	most
2	Better	21 C	5	Much worse	
3	about the same	1	6	First ski trip	

39. Do you plan to return to Winter Park for a ski trip next year?1 Yes3 Don't know

2 No

re you married? 1 Yes 2 No wo old are you? 1 Under 12 6 30-39 2 13-17 7 40-49 3 18-19 8 50-59 4 20-24 9 60 or over 5 25-29 0 No answer hat is the annual income of your household? 1 \$0-56,999 6 \$25,000-\$34,999 2 \$7,000-\$9,999 7 \$35,000-\$49,999 3 \$10,000-\$14,999 8 \$50,000-\$74,999 4 \$15,000-\$14,999 9 \$75,000 and over 5 \$20,000-\$24,999 9 \$75,000 and over 1 0-8th grade 5 College graduate 2 Some high school 6 Post-graduate work 3 High school graduate 7 Advanced degree 4 Some college 8 No answer hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student 5 Government or military 0 Other to you have any comments about Winter Park which would help make the area better place to ski? To e you to be willing to participate in similar additional questionnaires of his type sent to your home by mail? 1 Yes - 'go to 47a 2 No	1	Male	2	Female
1 Yes 2 No bw old are you? 6 30-39 1 Under 12 6 30-39 2 13-17 7 40-49 3 18-19 8 50-59 4 20-24 9 60 or over 5 25-29 0 No answer hat is the annual income of your household? 1 \$0-\$6,999 1 \$0-\$6,999 6 \$25,000-\$34,999 2 \$17,000-\$9,999 7 \$35,000-\$49,999 3 \$10,000-\$14,999 8 \$50,000-\$49,999 3 \$10,000-\$14,999 8 \$50,000-\$49,999 4 \$15,000-\$19,999 9 \$75,000 and over 5 \$20,000-\$24,999 0 No answer hat is the highest grade you have completed? 1 0-8th grade 1 0-8th grade 5 College graduate 2 Some high school 6 Post-graduate work 3 High school graduate 7 Advanced degree 4 Some college 8 No answer hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student 5 Government or military	-		-	
bw old are you? 1 Under 12 6 30-39 2 13-17 7 40-49 3 18-19 8 50-59 4 20-24 9 0 No answer hat is the annual income of your household? 1 \$0-\$6,999 6 \$25,000-\$34,999 1 \$0-\$6,999 6 \$25,000-\$74,999 3 \$10,000-\$14,999 8 \$50,000-\$74,999 3 \$10,000-\$14,999 8 \$50,000-\$74,999 3 \$52,000-\$24,999 0 No answer hat is the highest grade you have completed? 1 0-8th grade 5 College graduate 2 Some high school 6 Post-graduate work 3 High school graduate 7 Advanced degree 4 Some college 8 No answer 8 No answer hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker <	re you	married?		
1 Under 12 6 30-39 2 13-17 7 40-49 3 18-19 8 50-59 4 20-24 9 60 or over 5 25-29 0 No answer hat is the annual income of your household? 1 \$0-\$6,999 6 \$25,000-\$34,999 1 \$0-\$6,999 7 \$35,000-\$74,999 \$50,000-\$74,999 3 \$18,000-\$14,999 8 \$50,000-\$74,999 4 \$15,000-\$19,999 9 \$75,000 and over 5 \$20,000-\$24,999 0 No answer hat is the highest grade you have completed? 1 0-8th grade 5 2 Some high school 6 Post-graduate work 3 High school graduate 7 Advanced degree 4 Some college 8 No answer hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3	1	Yes	2	No
2 13-17 7 40-49 3 18-19 8 50-59 4 20-24 9 60 or over 5 25-29 0 No answer hat is the annual income of your household? 1 \$0-\$6,999 6 \$25,000-\$34,999 1 \$0-\$6,999 6 \$25,000-\$34,999 \$35,000-\$49,999 \$3\$10,000-\$14,999 8 \$50,000-\$74,999 3 \$10,000-\$14,999 8 \$50,000-\$74,999 \$5\$20,000-\$24,999 0 No answer hat is the highest grade you have completed? 1 0-8th grade 5 College graduate 1 0-8th grade 5 College graduate 7 Advanced degree 4 Some college 8 No answer 8 No hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student	ow old	are you?		
3 18-19 8 50-59 4 20-24 9 60 or over 5 25-29 0 No answer hat is the annual income of your household? 1 \$0-\$6,999 6 \$25,000-\$34,999 1 \$0-\$6,999 7 \$33,000-\$49,999 7 \$33,000-\$49,999 3 \$10,000-\$14,999 8 \$50,000-\$74,999 \$75,000 and over 5 \$20,000-\$24,999 0 No answer hat is the highest grade you have completed? 1 0-8th grade 5 College graduate 2 Some high school 6 Post-graduate work 3 High school graduate 7 Advanced degree 4 Some college 8 No answer No No No hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student 5 <td>1</td> <td></td> <td>-</td> <td></td>	1		-	
4 20-24 9 60 or over 5 25-29 0 No answer hat is the annual income of your household? 1 \$0-\$6,999 6 \$25,000-\$34,999 1 \$0-\$6,999 6 \$25,000-\$34,999 \$\$50,000-\$74,999 2 \$\$7,000-\$14,999 8 \$\$50,000-\$74,999 4 \$\$15,000-\$14,999 \$\$50,000-\$74,999 4 \$\$15,000-\$14,999 \$\$50,000-\$74,999 4 \$\$15,000-\$24,999 0 No answer hat is the highest grade you have completed? 1 0-8th grade 5 College graduate 1 0-8th grade 5 College graduate 7 Advanced degree 4 Some college 8 No answer No hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 5 Government or military 0 Other				
5 25-29 0 No answer hat is the annual income of your household? 1 \$0-\$6,999 6 \$25,000-\$34,999 2 \$7,000-\$9,999 7 \$35,000-\$49,999 3 \$10,000-\$14,999 8 \$50,000-\$49,999 3 \$10,000-\$14,999 8 \$50,000-\$49,999 9 \$75,000 and over 5 \$20,000-\$24,999 0 No answer hat is the highest grade you have completed? 1 0-8th grade 5 College graduate 2 Some high school 6 Post-graduate work 3 High school graduate 7 Advanced degree 4 Some college 8 No answer hat is your occupation? (If in doubt on classification, write in occupation) 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student 5 Government or military 0 Other better place to ski? <t< td=""><td></td><td></td><td></td><td></td></t<>				
hat is the annual income of your household? 1 \$0-\$6,999 6 \$25,000-\$34,999 2 \$7,000-\$9,999 7 \$35,000-\$49,999 3 \$10,000-\$14,999 8 \$50,000-\$74,999 4 \$15,000-\$19,999 9 \$75,000 and over 5 \$20,000-\$24,999 0 No answer hat is the highest grade you have completed? 1 0-8th grade 5 College graduate 2 Some high school 6 Post-graduate work 3 High school graduate 7 Advanced degree 4 Some college 8 No answer hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student 5 Government or military 0 Other o you have any comments about Winter Park which would help make the area better place to ski? 1 Yes - 'go to 47a 2 No 17. Name			. 9	60 or over
1 \$0-\$6,999 6 \$25,000-\$34,999 2 \$7,000-\$9,999 7 \$35,000-\$49,999 3 \$10,000-\$14,999 8 \$50,000-\$74,999 4 \$15,000-\$24,999 9 \$75,000 and over 5 \$20,000-\$24,999 0 No answer hat is the highest grade you have completed? 1 0-8th grade 5 College graduate work 3 High school graduate 7 Advanced degree 4 Some college 8 No answer hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student 5 Government or military 0 Other	5	25–29	0	No answer
<pre>2 \$7,000-\$9,999 7 \$35,000-\$49,999 3 \$10,000-\$14,999 8 \$50,000-\$74,999 4 \$15,000-\$24,999 9 \$75,000 and over 5 \$20,000-\$24,999 0 No answer hat is the highest grade you have completed? 1 0-8th grade 5 College graduate 2 Some high school 6 Post-graduate work 3 High school graduate 7 Advanced degree 4 Some college 8 No answer hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student 5 Government or military 0 Other</pre>	hat is	the annual income of your	household?	
3 \$10,000-\$14,999 8 \$50,000-\$74,999 4 \$15,000-\$19,999 9 \$75,000 and over 5 \$20,000-\$24,999 0 No answer hat is the highest grade you have completed? 1 0-8th grade 5 College graduate 1 0-8th grade 5 College graduate 7 Advanced degree 4 Some high school 6 Post-graduate work 3 High school graduate 7 Advanced degree 4 Some college 8 No answer 8 No enswer hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student 5 Government or military 0 Other 5 oyou have any comments about Winter Park which would help make the area better place to ski? 1 Yes - 'go to 47a 2 No 7a	1	\$0-\$6,999	6	\$25,000-\$34,999
4 \$15,000-\$19,999 9 \$75,000 and over 5 \$20,000-\$24,999 0 No answer hat is the highest grade you have completed? 1 0-8th grade 5 College graduate 2 Some high school 6 Post-graduate work 3 High school graduate 7 Advanced degree 4 Some college 8 No answer hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student 5 Government or military 0 Other o you have any comments about Winter Park which would help make the area better place to ski? 1 Yes - 'go to 47a 2 No				
5 \$20,000-\$24,999 0 No answer hat is the highest grade you have completed? 1 0-8th grade 5 College graduate 2 Some high school 6 Post-graduate work 3 High school graduate 7 Advanced degree 4 Some college 8 No answer hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student 5 Government or military 0 Other 5 Government or military 0 Other 6 better place to ski? 1 Yes - 'go to 47a 1 Yes - 'go to 47a 2 No				
hat is the highest grade you have completed? 1 0-8th grade 5 College graduate 2 Some high school 6 Post-graduate work 3 High school graduate 7 Advanced degree 4 Some college 8 No answer hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student 5 Government or military 0 Other o you have any comments about Winter Park which would help make the area better place to ski? Found you be willing to participate in similar additional questionnaires of his type sent to your home by mail? 1 Yes - 'go to 47a 2 No			9	-
1 0-8th grade 5 College graduate 2 Some high school 6 Post-graduate work 3 High school graduate 7 Advanced degree 4 Some college 8 No answer hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student 5 Government or military 0 Other o you have any comments about Winter Park which would help make the area better place to ski? ////////////////////////////////////	5	\$20,000-\$24,999	0	No answer
2 Some high school 6 Post-graduate work 3 High school graduate 7 Advanced degree 4 Some college 8 No answer hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 1 Executive, manager 6 Secretary, office work 7 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student 5 Government or military 0 Other 0 you have any comments about Winter Park which would help make the area 0 you have any comments about Winter Park which would help make the area 0 you be willing to participate in similar additional questionnaires of 1 Yes - 'go to 47a 2 No			completed?	
3 High school graduate 7 Advanced degree 4 Some college 8 No answer hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student 5 Government or military 0 Other o you have any comments about Winter Park which would help make the area better place to ski? 4 5 6 9 9 9 9 9 9 9 9 9 9 9 9 9 1 1 1 1 1 1 1 1 1 1 1 1 10 10 11 12		-		
4 Some college 8 No answer hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student 5 Government or military 0 Other				
hat is your occupation? (If in doubt on classification, write in occupation 1 Executive, manager 6 Secretary, office work 2 Professional, technical 7 Crafts, factory, mechanic 3 Teacher, professor 8 Homemaker 4 Salesman, buyer 9 Student 5 Government or military 0 Other to you have any comments about Winter Park which would help make the area better place to ski? You be willing to participate in similar additional questionnaires of this type sent to your home by mail? 1 Yes - 'go to 47a 2 No				
5 Government or military 0 Other o you have any comments about Winter Park which would help make the area better place to ski?			Berth, M.	그럼 이 문방 문화 말 수
better place to ski? Yould you be willing to participate in similar additional questionnaires of this type sent to your home by mail? 1 Yes - 'go to 47a 2 No 27a. Name	1 2 3	Executive, manager Professional, technical Teacher, professor	6 7 8	Secretary, office work Crafts, factory, mechanic Homemaker
this type sent to your home by mail? 2 No 1 Yes - 'go to 47a 2 No 7a. Name	1 2 3 4	Executive, manager Professional, technical Teacher, professor Salesman, buyer	6 7 8 9	Secretary, office work Crafts, factory, mechanic Homemaker Student
this type sent to your home by mail? 2 No 1 Yes - 'go to 47a 2 No 7a. Name	1 2 3 4 5 Do you	Executive, manager Professional, technical Teacher, professor Salesman, buyer Government or military have any comments about Wir	6 7 8 9 0	Secretary, office work Crafts, factory, mechanic Homemaker Student Other
this type sent to your home by mail? 2 No 1 Yes - 'go to 47a 2 No 7a. Name	1 2 3 4 5 00 you	Executive, manager Professional, technical Teacher, professor Salesman, buyer Government or military have any comments about Wir	6 7 8 9 0	Secretary, office work Crafts, factory, mechanic Homemaker Student Other
7a. Name	1 2 3 4 5 00 you	Executive, manager Professional, technical Teacher, professor Salesman, buyer Government or military have any comments about Wir	6 7 8 9 0	Secretary, office work Crafts, factory, mechanic Homemaker Student Other
	1 2 3 4 5 Do you a betto Would y	Executive, manager Professional, technical Teacher, professor Salesman, buyer Government or military have any comments about Wir er place to ski?	6 7 8 9 0 nter Park which te in similar 11?	Secretary, office work Crafts, factory, mechanic Homemaker Student Other ch would help make the area additional questionnaires of
	1 2 3 4 5 00 you a bette Nould y this ty 1	Executive, manager Professional, technical Teacher, professor Salesman, buyer Government or military have any comments about Win er place to ski? you be willing to participat ype sent to your home by man Yes - 'go to 47a	6 7 8 9 0 nter Park which te in similar 11? 2	Secretary, office work Crafts, factory, mechanic Homemaker Student Other ch would help make the area additional questionnaires of
City State Zip	1 2 3 4 5 Do you a bette Would y this ty 1 47a. 1	Executive, manager Professional, technical Teacher, professor Salesman, buyer Government or military have any comments about Wir er place to ski? you be willing to participat ype sent to your home by man Yes - 'go to 47a Name	6 7 8 9 0 nter Park which te in similar 11? 2	Secretary, office work Crafts, factory, mechanic Homemaker Student Other ch would help make the area additional questionnaires of No

•

1

.

Dear Copper Mountain Skier:

In an effort to provide our skiers with the best possible skiing experience, and to continually upgrade our facilities and service, we would appreciste you taking a few moments to complete a brief questionnaire. The survey consists of two parts:

> 1) This short questionniare (which we would like you to complete now and return to the ticket window).

and 2) A follow-up questionnaire to be mailed to you after you return home.

We would like to thank you in advance for your time and cooperation in helping us. Please accept this complimentary lift ticket for today as a token of our appreciation.

Again, thank you and have a good day:

COPPER MOUNTAIN, INC.

No. Part I - Survey 8. How many years have you been skiing? 1. What is your name and address? 1. [] 1 (first year) 5. [] 5 to 8 name 6. [] 9 to 12 2. [] 2. address 7. [] Over 12 · 3. [1 3 city state zip 4. [] 4 9. How do you classify yourself as a skier? 2. How old are you? (check one) 1. [] Under 12 4. [] 25-29 7. [] 40-49 1. [] Beginner 6. [] Advanced 5. [] 30-34 8. [] 50-59 2. [] 13-18 2. [] Novice 7. [] Expert 3. [] 19-24 6. [] 35-39 9. [] Over 59 3. [] Low Inter-mediate 8. [] Racer 3. What is your sex? (check one) 4. [] Intermediate 9. [] Freestyler 2. [] Female 1. [] Male 5. [] Advanced Intermediate 4. Are you married? (check one) 10. Are you on a ski vacation? (Ski Vacation = 3 or more nights) (check one) 2. [] No 1. [] Yes 5. What is your occupation? (check one) 2. [] No 1. [] Yes 6. [] Secretary, office work
 7. [] Crafts, factory, mechanic 1. [] Executive 11. If so, will you also be skiing any of 2. [] Professional control the following areas on this trip? (check as many as applicable) 8. [] Homemaker 3. [] Teacher, educator 1. [] A-Basin 6. [] Steamboat 9. [] Student 4. [] Salesman, buyer 7. [] Vail 2. [] Aspen 5. [] Government, military 10. Other 3. [] Breckenridge 8. [] Winter Park 6. How many people are in your party on 4. [] Keystone 9. [] Other this trip? 5. [] Loveland 7. How many people in your party are non-skiers? 12. Is this your first trip to Copper Mountain? (check one) Please return to ticket window 1. [] Yes 2. [] No THANK YOU COPPER MOUNTAIN, INC.

Sk1 Mountain Copper 31 De

4 10 ontin you sible very eas ted 10 r T 30 U 0 0 . uo 60 1 20 0 Ū BC 20 5 part onlv Ö QED ũ U 0 0 ao 0 202 C HO Vev 00 I II thi C give 0 0 not R đ obe skii 800 ŏ q o

participate tter 0 ٦ 00 0 Bon 5 IJ 0 1 fold tape Bk **T**b • ease 978-79 0 9 P1 Je la de 4 Ę Vev 0 10 0 Ð 8 ð > œ out シェリ L σ ack the complete 10 e r mail. 0 have dav U VC you ald < prep After 61 0 Mountain 0 the drawin t T T W Copper 0 ð and c 18

MOUNTAIN, COPPER

you

Thank

INC.

7 minutes Average completion time:

Part II - Mail	Burwy
1. What is your name and address?	
name	
address	
alter atota	
city state 2. How old are you? (cherk one)	zip
and the second se] 30-34 7. [] 40-49 9. [] 60 or
2. [] 13-18 4 [] 25-29 6. [ever
3. What is your sex? (check one)	
1. [] Hale 2. [] Female	
4. Are you married? (check one)	
1.[]Yes 2.[] No	
5. What is the highest level of educ	ation you have completed? (check one).
1. [] 0-8th	5. [] College graduate
2. [] Some high school	6. [] Post graduate work
3. [] High school graduate	7. [] Advanced degree
4. [] Some college/trade school	
6. What is the annual income of your	
1. [] 0-6,999 4. [] 15,	010-19,999 7. [] 35,000-49,999 000-24,999 8. [] 50,000-74,999
	000-34,999 9. [] 75,000 or over
 3. [] 10,000-14,999 6. [] 23. 7. What level of trail do you ski mo 	
	3. [] Most difficult (black diamon
2. [] Hore difficult (blue square	
8. How did you get to Copper Nountai	
1. [] Live at Copper Mountain	5. [] Airport limousine
2. [] Ski The Summit bus	6. [] Chartered bus
3. [] Private car	7. [] Scheduled bus
4. [] Rental car	8. [] Other
9. With whom did you come to Copper	Mountain? (check one)
1. [] Alone	5. [] Family & friends
2. [] Husband/wife only	6. [] Priends
3. [] Husband/wife & friends	7. [] Organized group/team
4. [] Family	8. [] Other
10. Had you ever skied at Copper Moun	tain before your trip? (check one)
1. [] Yes 2. [] No	,
11 If yes, how many days in total? (
1. [] 1 3. [] 3 5. [] 5 7. []	
2. [] 2 4. [] 4 6. [] 6 8. []	
 Did you ski at Copper Mountain wh (check one) 	ile on a convention trip to Colorado?
	If yes, what organization
I. [] 108 2. [] 30	
1. [] Yes 2. [] No 13. Are you a member of an organized	

No

75 14. Where do you ski most often? (check one) 1. [] A-Basin 9. [] Winter Park 5. [] Keystone 2. [] Aspen 6. [] Loveland 10. [] Other 3. [] Breckenridge 7. [] Steamboat 4. [] Copper Mountain 8. [] Vail From the following list, please indicate your 1st, 2nd and 3rd choice of ski areas in each category. (Please use the ski area number, not the name). 1. A-Basin 4. Copper Hountain 7. Steamboat 10. Other 2. Aspen 5. Levatone 8. Vail 3. Breckenridge 6. Loveland 9. Winter Park 1st choice 2nd choice 3rd choice . (16) (17) Ski Terrain (15) Mountain/Snov (19) (20) (15) (22) Base Facilities (21) (23) Friendliness (24) (25) (26) Ski School (27) (28) (29) (31) (32) Food Service (30) (34) (33) (35) Atmosphere (36) (37) (39) Total Experience 39. How many days did you ski during the 1976-77 ski season? 7. [] 15 - 21 4. [] 6 - 7 1. [] Mone 5. [] 22 - 28 2. [] 3 or less 5. 17 8 - 10 6. [] 11 - 14 3. [] 4 - 5 9. [] 29 or more 40. How many ski weekends did you take last year? (Ski Weekend = 1-2 mights) (Please check one) 4. [] 3 7. [] 7 to 8 1. [] None 2. [] 1 5. [] 4 8. [] 9 to 12 6. [] 5 to 6 9. [] Over 12 3. [] 2 41. How many ski vacations did you take last year? (Ski Vacation = 3 or more consecutive nights) - (Please check one) 7. [] 7 to 8 1. [] None 4. [] 3 5. [] 4 8. [] 9 to 12 2. [] 1 9 [] Over 12 6 [] 5 to 6 3. [] 2 42. How many nights did you spend away from home last year on skiing trips? (check one) 7. [] 15 - 21 4. [] 6 - 7 1. [] None 8. [] 22 - 28 5. [] 8 - 10 2. [] 3 or less 6. [] 11 - 14 9. [] 29 or more 3. [] 4 - 5 ... 43. How late in the season do you normally stop skiing? (check one) 3. [] April 5. [] June 1. [] February 4. [] Hay 2. [] March 44. How did you first hear about Copper Mountain? (check one) 7. [] Radio advertisement 1. [] Brochures 8. [] Travel agent 2. [] Ski shop employee 3. [] Magazine advertisement 9. [] Ski show 4. [] Newspaper advertisement 10. [] Other_ 5. [] Friend/word of mouth 11. [] Don't recall 12. [] Newspaper article 6. [] TV advertisement 13. [] Magazine article

which of the following do you read on a regular basis:

Athers as any as sportes					
45. [] Skit	56. 🚺 Lad	tes Home	Journal	67. []	Marioral Geographic
46. [] Skiing	57. [] New	Torker		68. []	Playboy
47. [] Texas Honthly	58. 门 Yev	week		69. []	
48. [] Southern Living	59. [] Wal	1 Street	Journal	70. []	U.S. Neve 6 World Report
4. [] Chicago Magazine	60. [] Eaq	uire		71. []	Glamour
50. [] Denver Magazine	61. [] Bus	iness We	ek	72. []	Mademoiselle
51. [] Colorado Magazine	62. [] Den	ver Post		73. []	Viva
S2. [] Powder	63. [] Roci	ky Mount	ain Nevs	74. []	People
53. [] Out	64. [] Fam	dly Circ	le	75. []	Vail Trail
54. [] Barrons	65 Pen	thouse		0	Others:
55. [] Aspen Times	66. 门 Spot	rts Illu	strated		
76. What radio station do					ae)
1. [] ROA	5. [] KOS		9.[] :	Other	
2. [] KBOW	6. 🗋 KDE			City	
3. [] KINS	7 KBP				×
4: [] ELAK	7. [] KLZ				
78. What types of after-ski		-			
77. [] Television	81 Jan				g/drinking
78. [] In-room movies	82 Din:	ing			tertainmat
79. [] Movie theater	68 Sho			Other	
80. [] Disco dancing	64 Con	versatio	3		
If available, would you pa (check as many as applicab	nticipate : ple)				
06. [] Might skiing			Bealth :		
80. [] Snowmobiling			Handball		tball
90. [] Sledding, toboggani	ng		Sleigh		
91. [] Indoor tennis		96. []	Night c	1058-COU	ntry skiing
98. [] Indoor swimping					
In which summer activities (check as many as applicable	i do you rej ile)	gularly	particip	ate?	
97. [] Tennis		ackpacki	ng/hiking	g 106. []	Boreeback
98. [] Golf	108.]] G	mping		108. []	Riding Speet
99. [] Sailing/water sport			climbing	g 107. []	Shooting Softball
00. [] Fishing/hunting					Bicycling
Please rate the performance Houstain. (circle the numb	e of the fo	ollowing	service	Dersonn	
					No Opinion/
100 7440	Excel.				No Contact
09. Lift attendants	1	2	3 4		6
10. Ski Patrol	1	2	3 4		6
11. Ticket sellers	1	2	3 4		6
18 Hental shop attendants		2	3 4	5	6
18. Center restaurant	1	2	3 4	5	ê
1% Solitude Station	i	2	3 4	5	6
ab. Mursery personnel	1	2	3 4	5	6
16. Sundry item sales personnel	1	2	3 4	5	6
17. Parking lot attendents					
tor accendince	1	2	3 4	5	6
 Parking lot attendants Bousekeeping personnel Rental office personne 	1	2 2	3 4 3 4	5	6 6

3

4 5

٠ 2 6

120. Bus drivers

What is your opinion of each of the following at Copper Nountain? (circle the appropriate number)

							He Opision,
		Excel.	Good	Avg.	Pair	Poor	
121	Lift line length	1	2	3	4	5	6
	Center Restaurant food	1	2	3	4	5	6
	. Solitude Station	1	2	3	4	5	6
194	Restaurant food Rental equipment	1	2	3	4	5	6
	Nursery program	1	2	3	4	s	6
125	Quality of skiing	1	2	3	4	5	6
	terrain . Slope grooming	1	2	3	4	5 .	- 6
	Lodging accoundations	1	2	3	4	. 5	6 .
	Lodging check-in	1	2	3	4	5	6
150	Reservation system	1	2	3	4	5	6
181	Trail working	. 1	2	3	4	5	6
	Village directional	1	2	3 -	4	5	6
183	signs Restroom cleanliness	1	2	3	4	5	6
	Other	1	2	3	4	5	6
184	Did you take a ski lesso	n at Coppe	r Nous	tain?	(checi	appro	priate box)
		1 50					
	If yes, what is you opin	-					
		Escel.	+	-			No Opinion
1.88	Trails used	1	2	3	4	5	6
	Classes offered	1	2	3	4	- 5	6
	Instructors	1	2	3		5	6
	Class size	1	2			5	6.
			2	3	1	5	. 6
180.	Instruction technique	1	2	3	4 trol at	5	6
180.		1	2	3	4 trol a	5	6
180.	Instruction technique Have you ever had any con (check appropriate box)	1	2 the S If no, for wh	3 Ski Pat	o quest	5 Coppe	6
189.	Instruction technique Have you ever had any cor (check appropriate box) 1. [] Tes. 2. [1 ntact with] No -	2 the S If no, for wh applic	3 Ski Pat	o quest	5 Coppe	6 er Nountain? 48. If yes,
140.	Instruction technique Have you ever had any con (check appropriate box) 1. [] Tes. 2. [[] Pirst aid (yourself or	1 ntact with] No - r a friend	2 the S If no, for wh applic	3 Ski Pat	o quest	5 Coppe	6 er Nountain? 48. If yes,
140. 141.	Instruction technique Have you ever had any cor (check appropriate box) 1. [] Tes. 2. [] [] Pirst aid (yourself or [] To ask for information	1 ntact with } No - r a friend	2 If no, for wh applic	3 Ski Pat , go to sat rea sable)	quest ison?	5 t Coppe tiom 1 (check	6 er Nountain? 48. If yes,
140. 141. 142.	Instruction technique Have you ever had any cor (check appropriate box) 1. [] Tes. 2. [] [] Pirst aid (yourself or [] To ask for information [] Warning (closure violation)	1 ntact with] No - r a friend a ation, ski	2 If no, for wh applic))	3 Ski Pat at rea sable)	quest ison?	5 t Coppe tiom 1 (check	6 er Hountain? 48. If yes,
189. 140. 141. 142. 145.	Instruction technique Have you ever had any cor (check appropriate box) 1. [] Tes 2. [[] Pirst aid (yourself or [] To sek for information [] Warning (closure viola [] Patrolman offered assisted	1 ntact with] No - r a friend a ation, ski	2 If no, for wh applic))	3 Ski Pat at rea sable)	quest ison?	5 t Coppe tiom 1 (check	6 er Hountain? 48. If yes,
189. 140. 141. 142. 145.	Instruction technique Have you ever had any cor (check appropriate box) 1. [] Tes 2. [] Pirst aid (yourself or [] To ask for information [] Warning (closure viola [] Patrolumn offered ass: [] Social greating	1 ntact with] No - r a friend a ation, ski	2 If no, for wh applic))	3 Ski Pat at rea sable)	quest ison?	5 t Coppe tiom 1 (check	6 er Hountain? 48. If yes,
149. 140. 141. 142. 145. 144.	Instruction technique Have you ever had any cor (check appropriate box) 1. [] Tes 2. [] Pirst aid (yourself or [] To ask for information [] Warning (closure viola [] Patrolmen offered ass: [] Social greeting [] Other	1 ntact with] No - r a friend a stion, ski istance af	2 If no, for wh applic)) ing to ter I	3 Ski Pat sat reusable) to fast fell.	t, etc.	5 t Coppe tiom 1 (check	6 er Nountain? 48. If yes,
149. 140. 141. 142. 145. 144.	Instruction technique Have you ever had any cor (check appropriate box) 1. [] Tes. 2. [] [] Pirst aid (yourself or [] To ask for information [] Warning (closure viola [] Patrolman offered ass: [] Social greeting [] Other Do you cross-country skill	1 ntact with] No - r a friend a stion, ski istance af ? (cbeck a	2 if no, for wh applic)) ing to ter I ppropr	3 Ski Par at rea cable) to fast fell.	t, etc.	5 t Coppe tiom 1 (check	6 Fr Nountain? 48. If yes, as miny as
140. 140. 141. 142. 145. 145.	Instruction technique Have you ever had any cor (check appropriate box) 1. [] Tes. 2. [] [] Pirst aid (yourself or [] To ask for information [] Warning (closure viola [] Patrolman offered ass: [] Social greating [] Other Do you cross-country skil 1. [] Yes 2. []	1 ntact with] No - r a friend a stion, ski istance af ? (check a] No	2 i the S if no, for wh applic) ing to ter I ppropr If no,	3 Ski Pal ski Pal	o quest son? t, etc. box) o quest	5 t Coppe tion 1 (check	6 fr Nountain? 40. If yes, as many as
140. 140. 141. 142. 145. 145.	Instruction technique Have you ever had any cor (check appropriate box) 1. [] Tes. 2. [] [] Pirst aid (yourself or [] To ask for information [] Warning (closure viola [] Patrolman offered ass: [] Social greeting [] Other Do you cross-country skill	1 ntact with] No - r a friend a stion, ski istance af ? (check a] No 1 while at] No	2 the fee start of the start of	3 Ski Pat sable) so fast fell. riate h ga to s your s your	<pre>> quest soon? (t, etc.) oon) > quest otain? </pre>	5 t Coppution 1 (check .))	6 er Nountain? 40. If yes, as many as 80. t appropriate box 50. If yes/ - the following?
189. 140. 141. 142. 145. 145.	Instruction technique Have you ever had any cor (check appropriate box) 1. [] Tes. 2. [] Pirst aid (yourself or [] To ask for information [] Warning (closure viola [] Patrolmen offered ass: [] Social greating [] Other Do you cross-country ski 1. [] Yes 2. [] Did you cross-country ski	1 ntact with] No - r a friend a stion, ski istance af ? (cbeck a] No 1 while at] No	2 i the S if no, for wh applic)) ing to ter I if no, Coppe If no, what i (circl	3 ski Pat sat res sable) so fast fell. fate b so to s your e apput	<pre>> quest sson? (t, etc.) oox) > quest stain? > quest opini opini</pre>	5 t Copput tion 1 (check)) (check tion 1 (checi tion 1 (checi tion 1 (checi	6 er Nountain? 40. If yes, as many as 80. t appropriate box 50. If yes/ - the following?
140. 141. 142. 145. 145. 146.	Instruction technique Have you ever had any cor (check appropriate box) 1. [] Tes. 2. [] Pirst aid (yourself or [] To ask for information [] Warning (closure viola [] Patrolmen offered ass: [] Social greating [] Other Do you cross-country ski 1. [] Yes 2. [] Did you cross-country ski	1 ntact with] No - r a friend a stion, ski istance af ? (cbeck a] No 1 while at] No	2 i the S if no, for wh applic)) ing to ter I if no, Coppe If no, what i (circl	3 ski Pat sat res sable) so fast fell. fate b so to s your e apput	<pre>> quest sson? (t, etc.) oox) > quest stain? > quest opini opini</pre>	5 t Copput tion 1 (check)) (check tion 1 (checi tion 1 (checi tion 1 (checi	6 er Nountain? 40. If yes, as many as 80. t appropriate box 50. If yes/ - the following? er)
140. 141. 142. 145. 145. 145. 145.	Instruction technique Have you ever had any cor (check appropriate box) 1. [] Tes. 2. [] [] Pirst aid (yourself or [] To ask for information [] To ask for information [] Warning (closure violation) [] Patrolman offered association [] Social greeting [] Other Do you cross-country ski 1. [] Yes 2. [] Did you cross-country ski 1. [] Yes 2. []	1 ntact with] No - r a friend a stion, ski istance af ? (check a] No i while at] No i while at] No	2 i the S if no, for whe applic ing to ter I ppropr If no, Coppe If no, what i (circl <u>Cood</u>	3 Ski Pai able) so fast fell. fate h ga to s your e appr <u>Avg.</u>	o quest soon? (t, etc. (t, etc. (t, etc. (t, etc. (t, etc.) (t, etc.)) (t, etc.)) ((5 i Coppution 1 (check)) ion 10 (check ion 10) (check ion 10) (check) (check) (check) (check) (check) (check) (check)	6 er Nountain? 48. If yes, as miny as 88. t appropriate box 38. If yes/ - the following? er) <u>No Opinion</u>
140. 140. 140. 141. 142. 145. 145. 146. 146. 146.	Instruction technique Have you ever had any cor (check appropriate box) 1. [] Tes. 2. [] [] Pirst aid (yourself or [] To ask for information [] Warning (closure viola [] Patrolmen offered ass: [] Social greating [] Other Do you cross-country ski 1. [] Yes 2. [] Did you cross-country ski 1. [] Yes 2; [] Traila	1 ntact with] No - r a friend a stion, ski istance af ? (check a] No i while at] No i while at] No 1	2 i the S if no, for wh applic)) ing to ter I ing to ter I ppropr If no, Coppe If no, what i (circl <u>Cood</u> 2	3 Ski Pal (so fast rable) oo fast fell. Tiate b go to a your e appr <u>Avg.</u> 3	o quest soon) o quest t, etc. oon) o quest opini copini <u>Fair</u> 4	5 copput tion 1 (check .) .) :ton 11 (checi ion 11 (checi ion 11 <u>for a unit</u> <u>Foor</u> 5	6 er Nountain? 48. If yes, as many as 88. t appropriate box 58. If yes/ - the following? er) <u>No Opinion</u> 6
141. 142. 143. 145. 145. 145. 145. 145. 145. 145. 145	Instruction technique Have you ever had any cor (check appropriate box) 1. [] Tes. 2. [] [] Pirst aid (yourself or [] To ask for information [] Warning (closure viola [] Patrolmen offered ass: [] Social greating [] Other Do you cross-country ski 1. [] Yes 2. [] Did you cross-country ski 1. [] Yes 2; [] Traila Instructors	1 ntact with] No - r a friend a stion, ski istance af ? (check a] No 1 while at] No <u>Excel.</u> 1 1	2 2 the S 1f no, for wh applic)) ing to ter I ppropr If no, what i (circl <u>Coope</u> If no, what i 2 2	3 3 3 3 3 3 3 3 3 3 3 3 3 3	o quess sson? (t, etc. oox) o quest stain? o quest stain? <u>Fair</u> 4 4	5 copposition 1 (check cion 1 (check cion 1 (check cion 1 (check cion 1 (check cion 1 cion	6 Fr Nountain? 48. If yes, as many as 88. 1 appropriate box 88. If yes/ - the following? er? <u>No Opinion</u> 6 6
141. 141. 142. 145. 145. 146. 147. 148. 149. 180. 151.	Instruction technique Have you ever had any cor (check appropriate box) 1. [] Tes. 2. [] [] Pirst aid (yourself or [] To ask for information [] Warning (closure violat [] Patrolmen offered ass: [] Social greating [] Other	1 ntact with] No - r a friend a tion, ski istance af ? (check a] No 1 while at] No 1 while at] No 1 min 1 1	2 2 2 2 2 2 2 2 2 2 2 2 2 2	3 Ski Pal (go to tat reveable) o fast fell. Tate h go to a your e appr Avg. 3 3 3	<pre>> quest sson? (t, etc.) opm stain? > quest stain? > quest stain? 4 4 4 4 4 4</pre>	5 copposition 1 (check .) clon 12 (check clon 12 (check clon 12 (check clon 12 (check clon 12 check clon 12 clon	6 Fr Nountain? 46. If yes, as many as 58. 1 appropriate box 58. If yes/ the following? er) No Opinion 6 6 6 6
141. 142. 144. 145. 145. 146. 146. 146. 147. 146. 146. 146. 146. 146. 146. 146. 146	Instruction technique Have you ever had any cor (check appropriate box) 1. [] Tes. 2. [] [] Pirst aid (yourself or [] To ask for information [] Warning (closure viola [] Patrolman offered ass: [] Social greating [] Other Do you cross-country skill 1. [] Yes 2. [] Did you cross-country skill 1. [] Yes 2. [] Did you cross-country skill 1. [] Yes 2. [] Traila Instructors Classes offered Class cize	1 ntact with] No - r s friend stion, shi istance af ? (check s] No 1 while ac] No 1 while ac] No 1 this ac	2 2 2 2 2 2 2 2 2 2 2 2 2 2	3 Ski Pau able) so fasi fell. riste k go to r Houm sp to a your 3 3 3 3 3	<pre>9 quest sson? / t, etc. (or)) quest opini pini f a 4 4 4 4 4 4 4</pre>	5 5 Copper cion 1 (check)) cion 11 (checi ion 11 (checi ion 11 (checi 5 5 5 5 5	6 Fr Nountain? 46. If yes, as many as 58. 1 appropriate box 58. If yes/ - the following? 50. 50. 50. 50. 50. 50. 50. 50.

Please rate the importance of each of the following possible reasons for skiing. (circle the appropriate number) 76

	Tery Important	Important	Of Some	/ D Of Little Importance	Unimporta
165. Development of skiing skills	1	2	3	4	5
186. Skiing to be with others	1	2	3	4	5
167. Skiing for ex- citement	1	2	3	. 4	5
188 Skiing for group acceptance	1	2	3	4	5
169. Skiing to relax	1	2	3	4	5
160. Skiing for ex- Vercise	1	2	3	4	5
Are you aware that Con	ner Nomintal	n has a tot	al Master P	lan for the	develop-

you aware that Coppen ment of: (check appropriate box)

161.	The	Houstain?	1. []	Yes	2.	Ð	No
162.	The	Village?	1. []	Yes	2.	[]	36

165. Did you ski Copper Mountain while on an overnight trip, or on a day trip! (check appropriate box)

1. [] Overnight 2. [] Day trip

Okay, here's your chance.

Tell us what you don't like about Copper Mountain or ski areas in general. (Additional space on back, if necessary.)

IF YOU SKIED COPPER MOUNTAIN WHILE ON VACATION (3 or more nights), PLEASE ANSWER QUESTIONS 164-190. (Even if you did not stay at Copper Mountain)

THANK YOU

24

(

VACATION SECTION stravel to Colorado?

TOF now are you traver to	COTOLEGO:			
1. [] Live in Colorado	4. [] Schedu	led bus	7. [] Private airpl	ane -
2. [] Private car	5. [] Charte	red bus	8. [] Chartered air	plane
3. [] Rental car	6. [] Comment	rial Airli	ne 9. [] Other	
165. Where did you stay whi	le skiing at (Copper Moun	tain?	
1. [] Copper Mountain	4. [] Silver	thorne	7. [] Vail	
2. [] Frisco	5. [] Keysto	me ·	8. [] Other-	
3. [] Dillon	6. [] Brecke	mridge	9. [] In transit to	other
166. How many nights did ;	Ame brene unu	-	Print this tring	
(check appropriate bos		9 11 018 1104	a on the try.	
1. [] None-	4. [] 3 ·		7. [] 6	
2. [] 1	5. [] 4 -		8. [] 7	
3. [] 2	6. [] 5		9. [] 8 or more	
167. If you stayed in Coppe				ou-
believe a car is neces		ppropriate	box)	
1. [] Yes	2. [] %0			
168. How far in advance did		*		
1. [] 1 mo. or less			. 7. [] Don't recall/	
2. [] 2-3 mos.	5. [] 7-12 e			14
	6. [] More t			
109. How far in advance did (check appropriate Sca	:)			
1. [] 1 mo. 'or less	4. [] 6 mos.		7. [] Den't recall/ No answer	
2. [] 2-3 mos.	5. [] 7-12 1	05.		
3. [] 4-5 mos.	6. [] More 1			
170. In planning your ski v (check appropriate box	acation, what	was your p	rimmry source of infor	mation?
1. [] Airline, which o	ne?	7, [] 9	ki shows	
2. [] Travel agent, wh			ki Club or other organ	isation
3. [] Copper Hountain literature	brochure/	9. [] :	lagazine article	
4. [] Magazine adverti	sement -	19. [] :	euspaper article	
5. [] Newspaper advert	isement	n. II e	ther	
6. [] Word of mouth				
171. Through whom did you m (check appropriate box		ion arrang	ements?	
1. [] Travel agent, wh	ich one?	5. [] s	ki club or other organ	ized
2. [] Airlines, which	one?	6. [] 1	group riend/relative	
3. [] Resort Associati		7. [] 0		
4. [] Direct to account				
In addition to Copper Mount		ki any of	the following areas on	Tour
vacation? (answer as many a				,
	Ho	w Many Day	<u>s</u> ?	1
172 [] A-Basin			13. Jac	
175. [] Breckenridge	1.			

	(check appropriate box)	5. [] Stayed with friends
	2. [] Condominium	b. [] Camper/RV
	3. [] Dormitory/hostel	7. [] Other
	4. [] Rented house	7. [] other
		eration when seeking accompositions?
	(check as many as applicable)	eration when seeking accompations:
	181 [] Fireplace	
	182. [] Sauna	
	185. [] Jaccuzzi	
	184. [] Building elevators	
	185. [] Ski slope view	
	186. [] Covered parking	
	187. [] Proximity to lifts	
	188. [] Proximity to restaurants	
	Other	
	110	
e 19 1		
		nsportation come to Copper Mountain
	1. Transportation cost?	
	1. Transportation cost?	ther expenses of your group
	Transportation cost? Tor how many people?	ther expenses of your group
	Transportation cost? Por how many people? 190. Please estimate the total of all of 1. Total expenses?	ther expenses of your group
	Transportation cost? Por how many people? 190. Please estimate the total of all of 1. Total expenses?	ther expenses of your group
	Transportation cost? Por how many people? 190. Please estimate the total of all of 1. Total expenses?	ther expenses of your group
	Transportation cost? Tor how many people? 190. Please estimate the total of all of 1. Total expenses? 2. For how many people?	ther expenses of your group
	Transportation cost? For how many people? 190. Please estimate the total of all of 1. Total expenses? 2. For how many people? Thank you	ther expenses of your group
	Transportation cost? For how many people? 190. Please estimate the total of all of 1. Total expenses? 2. For how many people? Thank you	ther expenses of your group
	Transportation cost? For how many people? 190. Please estimate the total of all of 1. Total expenses? 2. For how many people? Thank you	ther expenses of your group
	Transportation cost? Por how many people? 190. Please estimate the total of all of Total expenses? 2. For how many people? Thank you COPPER MOL	Aber expenses of your group
	I. Transportation cost? Por how many people? 190. Please estimate the total of all of 1. Total expenses? 2. For how many people? Thank you COPPER MOUNTAIN THANKS YOU FO	once again. MIAIN, INC. SR YOUR TIME AND COOPERATION. STURNED SURVEYS WILL BE ELIGIBLE TO 8-DAY LIFT PACKAGES FOR THE

			HOW HERY DAYS:
172	[]	A-Basin	
175.	[]	Breckenridge	
174.	[]	Aspen	
178.	[]	Leystone	
176.	[]	Loveland	
177.	[]	Steamboat	
178.	[]	Vail	
179.	[]	Winter Park	
		Ither	